On the detection of COVID-driven changes in atmospheric carbon dioxide

Nicole Suzanne Lovenduski¹, Abhishek Chatterjee², Neil C. Swart³, John Fyfe⁴, Ralph F. F. Keeling⁵, and David Schimel⁶

¹University of Colorado Boulder ²NASA Goddard Space Flight Center ³Environment Canada ⁴Canadian Centre for Climate Modelling and Analysis ⁵University of California, San Diego ⁶Jet Propulsion Laboratory

November 23, 2022

Abstract

We assess the detectability of COVID-like emissions reductions in global atmospheric CO_2 concentrations using a suite of large ensembles conducted with an Earth system model. We find a unique fingerprint of COVID in the simulated growth rate of CO_2 sampled at the locations of surface measurement sites. Negative anomalies in growth rates persist from January 2020 through December 2021, reaching a maximum in February 2021. However, this fingerprint is not formally detectable unless we force the model with unrealistically large emissions reductions. Internal variability and carbon-concentration feedbacks obscure the detectability of short-term emission reductions in atmospheric CO_2 . COVID-driven changes in the simulated interhemispheric CO_2 gradient and column-averaged dry air mole fractions of CO_2 (total column or XCO_2) are eclipsed by large internal variability. Carbon-concentration feedbacks begin to operate almost immediately after the emissions reduction; these feedbacks reduce the emissions-driven signal in the atmosphere carbon reservoir and further confound signal detection.

On the detection of COVID-driven changes in atmospheric carbon dioxide

Nicole S. Lovenduski¹, Abhishek Chatterjee ^{2,3}, Neil C. Swart⁴, John C. Fyfe⁴, Ralph F. Keeling⁵, David Schimel⁶

5	¹ Department of Atmospheric and Oceanic Sciences and Institute of Arctic and Alpine Research, University of Colorado,
6	Boulder, CO, USA
7	² Universities Space Research Association, Columbia, MD, USA
8	³ National Aeronautics and Space Administration, Goddard Space Flight Center, Greenbelt, MD, USA
9	⁴ Canadian Centre for Climate Modelling and Analysis, Environment and Climate Change Canada, Victoria, BC, Canada
10	⁵ Scripps Institution of Oceanography, University of California San Diego, La Jolla, CA, USA
11	⁶ Jet Propulsion Laboratory, California Institute of Technology, Pasadena, CA, USA

Key Points:

1

2

3 4

12

13	•	Climate model simulations suggest a lagged response in the growth rate of atmo-
14		spheric CO ₂ due to COVID-19 emissions reductions
15	•	Detection of this reduction in observations is hampered by internal variability com-
16		bined with reduced ocean and land uptake of CO ₂
17	•	Our results foreshadow the challenges of detecting the effects of CO ₂ mitigation ef-
18		forts to meet the Paris climate agreement

 $Corresponding \ author: \ Nicole \ S. \ Lovenduski, \ \texttt{nicole.lovenduski@colorado.edu}$

19 Abstract

- $_{20}$ We assess the detectability of COVID-like emissions reductions in global atmospheric CO₂
- concentrations using a suite of large ensembles conducted with an Earth system model. We
- find a unique fingerprint of COVID in the simulated growth rate of CO₂ sampled at the loca-
- tions of surface measurement sites. Negative anomalies in growth rates persist from January
- ²⁴ 2020 through December 2021, reaching a maximum in February 2021. However, this finger-
- print is not formally detectable unless we force the model with unrealistically large emissions
- reductions. Internal variability and carbon-concentration feedbacks obscure the detectability
- of short-term emission reductions in atmospheric CO_2 . COVID-driven changes in the simulated interhemispheric CO_2 gradient and column-averaged dry air mole fractions of CO_2
- ²⁸ ulated interhemispheric CO_2 gradient and column-averaged dry air mole fractions of CO_2 ²⁹ (total column or XCO_2) are eclipsed by large internal variability. Carbon-concentration feed-
- (total column or XCO₂) are eclipsed by large internal variability. Carbon-concentration feed backs begin to operate almost immediately after the emissions reduction; these feedbacks
- reduce the emissions-driven signal in the atmosphere carbon reservoir and further confound
- 32 signal detection.

³³ Plain Language Summary

COVID pandemic lockdowns suddenly slowed the rate at which we burned fossil fuels and 34 released carbon dioxide into the atmosphere, yet we cannot find any significant reductions 35 in the growth of carbon dioxide in the atmosphere from our measurements. Here we provide 36 some reasons to explain this conundrum. We use a climate model to mimic the changes in 37 atmospheric carbon that would occur with different amounts of reductions in fossil fuel burning. We find that it is hard to see the change in fossil fuel burning in atmospheric carbon or 39 its growth because of a large background component of natural variability. In addition, once 40 we reduce our fossil fuel burning and the amount of carbon dioxide in the atmosphere de-41 creases, the ocean and land also stop taking up as much carbon as normal. As we will soon 42 lower our fossil fuel burning on purpose to slow climate change, our findings forewarn of the 43 difficulties of detecting the effects of this in measurements of atmospheric carbon dioxide. 44

45 **1 Introduction**

Falling energy demand during the COVID-19 pandemic led to rapid decreases in energy-46 related carbon dioxide (CO₂) emissions. In 2020, global annual CO₂ emissions fell by 7% to 47 2011 levels (9.3 Pg C yr⁻¹), and the rapid decline in emissions during the first half of 2020 48 surpassed the rate of emission declines during any previous economic recession or World 49 War II [Le Quéré et al., 2020; Forster et al., 2020; Liu et al., 2020; Friedlingstein et al., 50 2020]. Global annual CO₂ emissions are forecast to remain below 2019 levels through 2021, 51 and subsequent recovery of emissions is expected within a few years [International Energy 52 Agency, 2021; Le Quéré et al., 2021]. The precipitous and short-lived drop in emissions dur-53 ing the COVID pandemic offers a unique opportunity to assess the detection of these types of 54 emissions declines in observations of the global carbon cycle. 55

While the COVID-related CO2 emissions reductions had a measurable impact on re-56 gional atmospheric CO₂ concentrations [Chevallier et al., 2020; Tohjima et al., 2020; Turner 57 et al., 2020; Buchwitz et al., 2021; Liu et al., 2021; Wu et al., 2021], as of this writing, there 58 is no indication of a global-scale decrease in the atmospheric CO_2 mixing ratio or its growth 59 rate due to the emissions reductions [World Meteorological Organization, 2020; NOAA Global 60 Monitoring Laboratory, 2021]. Even with a robust global measurement system, the de-61 tection of COVID-related emissions reductions in global CO₂ or its growth rate is chal-62 lenging due to two factors: (1) internal variability in the climate system, and (2) carbon-63 concentration feedbacks. Internal variability is unforced climate variability that arises from 64 the coupled interactions of the atmosphere and ocean [e.g., El Niño-Southern Oscillation (ENSO); *Deser et al.*, 2012a]. The role of internal variability in the growth rate of CO₂ has 66 been well documented in the literature [e.g., Keeling et al., 2001; Sarmiento and Gruber, 67 2002; Frölicher et al., 2013], and multiple studies implicate this variability in our inabil-68

ity to detect emissions changes in measurements of atmospheric CO₂ [e.g., Peters et al., 69 2017]. Carbon-concentration feedbacks manifest from the sensitivity of the ocean and land 70 carbon reservoirs to changing CO₂ [Friedlingstein et al., 2006; Arora et al., 2013, 2020]. 71 Recent studies suggest that the ocean carbon reservoir rapidly responds to perturbations in 72 CO₂ [*McKinley et al.*, 2020; *Ridge and McKinley*, 2021], and this can further confound the 73 detection of emissions changes in measurements of atmospheric CO_2 . It is critical that we 74 develop a deeper understanding of the role of internal variability and carbon feedbacks on 75 the detectability of emissions changes to inform both near-term (1-10 year) predictions of the 76 carbon cycle [Ilyina et al., 2021] and the verification of future emissions reductions [Peters 77 et al., 2017; Ridge and McKinley, 2021]. 78

Initial-condition large ensembles of Earth system models are a relatively new tool that 79 provide a means to quantify the anthropogenic influence on the Earth system in the presence 80 of internal climate variability [Deser et al., 2020]. These large ensembles are a set of simu-81 lations with a single Earth system model: each simulation or ensemble member is initialized 82 slightly differently to create diverging climate trajectories, while all ensemble members are 83 externally forced with a common emission scenario or radiative forcing prescription [Deser et al., 2012b]. Multiple studies have used large ensembles to account for the role of inter-85 nal variability in long-term climate trends [e.g., Deser et al., 2012a,b, 2016]. Most recently, 86 large ensembles have been used to estimate the anthropogenic influence on short-term cli-87 mate signals [such as for the COVID pandemic, see, e.g., Fyfe et al., 2021; Gettelman et al., 88 2021; Jones et al., 2021], and to make Earth system predictions over the near-term [1-10 89 years; Yeager et al., 2018]. However, no studies have used a large ensemble framework to 90 assess the detectability of short-term CO₂ emissions reductions from atmospheric CO₂ mea-91 surements. 92

Here, we develop an understanding of the role of internal variability and carbon feed-93 backs on the detectability of a short-lived CO_2 emissions reduction in the atmospheric mix-94 ing ratio of CO_2 (χCO_2) using output from an initial-condition large ensemble of an Earth 95 system model. This 30-member ensemble evolves the Earth system under three, short-term 96 emissions reduction scenarios of differing magnitudes. We investigate the detectability of the 97 emissions reduction using several modeled parameters that characterize atmospheric CO₂: 98 the interhemispheric χCO_2 difference, the χCO_2 growth rate, the column-averaged CO₂, and 99 the atmospheric carbon reservoir. 100

101 2 Methods

We utilize the Canadian Earth System Model version 5 (CanESM5), which consists of 102 coupled atmosphere, ocean, sea-ice, land surface, and land/ocean carbon cycle model com-103 ponents [Swart et al., 2019]. The atmospheric model in CanESM5 is version 5 of the Cana-104 dian Atmospheric Model (CanAM5) that has an approximate 2.8° horizontal resolution and 105 49 vertical levels of varying thickness on a hybrid sigma-pressure vertical grid, and similar 106 physical parameterizations as its predecessor [CanAM4; Swart et al., 2019]. The land com-107 ponent of CanESM5 consists of the Canadian Land Surface Scheme (CLASS) and the Cana-108 dian Terrestrial Ecosystem Model (CTEM) that produce fluxes of energy, water, and carbon 109 dioxide at the land-atmosphere interface via the simulation of physical and biogeochemical 110 processes, including the CO₂ fertilization of photosynthesis [Swart et al., 2019]. The ocean 111 physical and biogeochemical components of CanESM5 used in this study are the CanNEMO 112 physical model coupled to the Canadian Model of Ocean Carbon (CMOC), which simulates 113 ocean carbon and its exchange with the atmosphere at approximately 1° horizontal resolution 114 [Swart et al., 2019]. In our study, the concentration of CO_2 in the CanESM5 atmosphere is 115 modeled as a three-dimensional, prognostic passive tracer that responds to air-sea and air-116 land CO₂ fluxes from the coupled land and ocean carbon cycle components, and to specified 117 CO₂ emissions. 118



Figure 1. Annual-mean, de-trended interhemispheric difference in χ CO₂ (Mauna Loa minus South Pole; ppm) from (purple) observations, (gray/black) the CanESM5 historical simulations, and (red) the CanESM5 COVID-like simulations. Gray lines show individual model ensemble members, and thick black line shows the ensemble mean. Red dot and range illustrates the mean, maximum, and minimum interhemispheric difference in 2020 from the CanESM5 COVID-like ensemble. Numbers in parenthesis on legend correspond to the number of ensemble members plotted. Periods of marked emissions declines are indicated by dashed red vertical lines.

We analyze output from five ensembles of CanESM5. In each case, ensemble members 119 are initialized with slightly perturbed climate states to simulate a range of internal variabil-120 ity, but each member in a given ensemble experiences identical external forcing. The first 121 ensemble (the historical ensemble) covers the period from 1750 to 2014 and consists of 9 en-122 semble members of CanESM5 forced with a global historical emission data set of CO2 and 123 other climate-relevant gases and aerosols - this was devised for emissions-driven historical 124 simulations in Phase 6 of the Coupled Model Intercomparison Project [CMIP6; Figure S1a; 125 Hoesly et al., 2018]. The second ensemble (the control ensemble) covers the period 2015-126 2100 and consists of 30 ensemble members of CanESM5 integrated under the esm-SSP2-127 4.5 emissions scenario [Figure S1b; O'Neill et al., 2016]. The remaining 3 ensembles span 128 2019-2040 and consist of 30 members each that are forced with COVID-like CO2 emissions 129 reductions beginning in December 2019 and resolving in December 2021; peak emissions re-130 ductions of 25% (COVID-like), 50% (2 \times COVID-like), and 100% (4 \times COVID-like) occur 131 in May 2020 (Figure S1b). Hereafter, we refer to these later three ensembles collectively as 132 the CanESM5-COVID ensemble, as described in Fyfe et al. [2021] and Lovenduski et al. 133 [2021]. The CO_2 emissions in the historical and control ensembles have spatial and sea-134 sonal variability; emissions are highest near urban centers in the Northern Hemisphere (Fig-135 ure S2a) and peak in boreal winter when energy consumption in the Northern Hemisphere 136 is at a maximum (Figure S2b). Emissions are scaled uniformly for the COVID ensembles 137 to maintain this spatial and seasonal variability. In the CanESM5-COVID ensemble output 138 we analyze here, emissions of CO_2 from other sources (e.g., land use change) and emissions 139 of other climate relevant gases and aerosols are prescribed from the esm-SSP2-4.5 scenario, 140 i.e., these emissions do not change due to COVID. 141

The global carbon cycle in CanESM5 compares well with observational metrics and is thus an appropriate tool for the study of the detectability of short-term emissions reductions in atmospheric χ CO₂. Air-sea and air-land CO₂ fluxes from the historical simulation of CanESM5 were previously evaluated in *Swart et al.* [2019]. Briefly, *Swart et al.* [2019] illustrate high skill and low root mean square error between simulated and observed spatial pat-



Figure 2. Temporal evolution of the growth rate of de-seasoned, monthly χCO_2 from the CanESM5 170 COVID ensemble sampled at (a) Mauna Loa, and (b) the average of 12 flask sites [as in Cadule et al., 2010] 171 over 2020-2024. Growth rate is calculated as the difference in χCO_2 for a given month relative to the same 172 month in the previous year. Thin lines show individual ensemble members, and thick lines show the ensemble 173 mean for each emissions scenario. Red dot and range illustrates the mean and 2σ (95%) confidence interval in 174 February 2021 for the COVID-like emissions scenario. Subplots show the temporal correlation coefficients of 175 individual ensemble members with the ensemble mean over Jan 2020 - Dec 2021 for each emissions scenario. 176 Small circles show the correlation coefficients across the 30 ensemble members, large circles show the mean 177 correlation coefficients, and dashes indicate 2σ (95%) confidence intervals. 178

terns of Gross Primary Production (GPP) and air-sea CO₂ flux over 1981 to 2010. CanESM5 154 tends to overestimate GPP in sub-saharan Africa and underestimate GPP in the Amazon 155 rainforest, likely due to precipitation biases [Swart et al., 2019]. Historical CanESM5 air-156 sea CO₂ fluxes are biased high in the North Atlantic and low in the Southern Ocean, such 157 that the globally integrated air-sea CO₂ flux exhibits little bias as compared to observations [Swart et al., 2019]. CanESM5 captures the broad features of the amplitude and phasing of 159 the seasonal cycle of χCO_2 measured at Barrow (BRW), Mauna Loa (MLO), and South Pole 160 (SPO), though the seasonal drawdown of CO_2 occurs too early at Point Barrow, and the am-161 plitude is biased high at Mauna Loa (Figure S3; the model is sampled at the approximate 162 latitude, longitude, and height of the flask sample in the real world). Finally, the CanESM5 163 control ensemble mean exhibits a similar growth rate in χCO_2 (2.4 ppm yr⁻¹ over 2015-164 2019; see Figure 2) as calculated from observations $(2.57 \pm 0.08 \text{ ppm yr}^{-1} \text{ over } 2015\text{-}2019;$ 165 https://gml.noaa.gov/ccgg/trends/gl_gr.html). The actual growth rate derived from observa-166 tions is slightly higher due to the impact of the 2015-2016 El Niño event on the carbon cycle 167

¹⁶⁸ [*Chatterjee et al.*, 2017; *Liu et al.*, 2017].

169 **3 Results**

The de-trended interhemispheric gradient in observed, annual mean χCO_2 exhibits large annual-to-decadal fluctuations over 1960-2020 that are generally replicated by the model but have little correlation with past periods of marked emissions reductions (Figure 1). The interhemispheric gradient (here expressed as the interhemispheric difference, Mauna



Figure 3. Temporal evolution of monthly, column-averaged χ CO₂ over (a) the Northern Hemisphere,

 $_{198}$ 20°N-55°N, and (b) the Southern Hemisphere, 20°S-55°S, simulated with the CanESM5 COVID ensemble.

Thin lines show individual ensemble members, and thick lines show the ensemble mean for each emissions scenario.

Loa minus South Pole) can be a useful indicator of the sources and sinks of CO₂ [Dargav-183 ille et al., 2003], and its time-varying behavior can indicate changes in CO2 sources or sinks 184 [*Ciais et al.*, 2019], such as fossil fuel emissions. However, fluctuations in the observed inter-185 hemispheric difference display no correlation with past periods of emissions reductions, nor 186 with the ongoing emissions reductions due to COVID (Figure 1; emissions history in Fig-187 ure S1a). The de-trended interhemispheric difference in the CanESM5 historical ensemble 188 members encapsulate the observations and the ensemble mean replicates the decadal varia-189 tions in the observations, though the interannual variance of individual ensemble members 190 is greater than that of the observational record (Figure 1). The CanESM5 COVID-like en-191 semble mean simulates a negative anomaly in the interhemispheric difference in 2020, with 192 more than 50% of the ensemble members showing a negative anomaly (Figure 1). This is in 193 disagreement with the observational record, akin to a single ensemble member in the large 194 ensemble framework, for which we observe a small positive anomaly in the interhemispheric 195 difference in 2020 (Figure 1).

The 30-member CanESM5 COVID ensemble predicts a decrease in the de-seasoned, 201 monthly growth rate of χCO_2 from January 2020 through February 2021, followed by an 202 increase in growth rate from February through December 2021 under all of the COVID emis-203 sion scenarios when sampled at both Mauna Loa and 12 global flask sites [flask sites as in 204 *Cadule et al.*, 2010, Figure 2]. The decrease in the growth rate is largest for the $4 \times \text{COVID}$ -205 like emissions scenario and smallest for the COVID-like emissions scenario and peaks in 206 February 2021 under all COVID scenarios (Figure 2). Meanwhile, the control ensemble 207 exhibits little change in its growth rate over this period (Figure 2). This suggests that the 208 χCO_2 growth rate is highly sensitive to the magnitude of the emissions reduction and that 209 growth rate anomalies at Mauna Loa and across the global flask network tend to be largest 210 \sim 9 months after the peak emissions reduction (May 2020; Figure S1b). Under the 4 \times COVID-211 like scenario, the growth rate exceeds the control growth rate from 2022 through 2024 before 212 returning to control values (Figure 2). Figure 2 also reveals that internal variability tends to 213 obscure the emissions reduction signal in χCO_2 at an individual site more than in the global 214 average (cf. Figures 2a and 2b); averaging across multiple sites tends to dampen the effects 215 of internal variability that manifest most strongly at local and regional scales [Hawkins and 216



Figure 4. Cumulative changes in the (a) atmosphere, (b) ocean, and (c) land carbon reservoirs from December 2019 onwards, as simulated by the CanESM5 COVID ensemble. Colored lines show the anomaly in the ensemble-mean reservoir size relative to the control ensemble mean (SSP2-4.5), and gray shading indicates the spread in the cumulative reservoir anomaly across the control ensemble. Dashed lines in (a) show the cumulative changes in atmospheric carbon due to anomalous emissions alone.

Sutton, 2009]. As a result, the ensemble-mean February 2021 COVID-like χ CO₂ growth rate is significantly different from the control ensemble mean in the average of the 12 flask sites, but not at Mauna Loa (Figure 2; significance calculated using a 2σ (95%) confidence interval across the COVID-like ensemble members). If we wish to detect a signal of the COVIDdriven emissions reduction in the real-world growth rate of χ CO₂, our modeling study suggests that we are most likely to find it in early 2021 by averaging across measurements collected in the global flask network.

Is it possible to detect the change in the de-seasoned, monthly χCO_2 growth rate from 229 flask observations in the real world, where we have only a single "ensemble member"? To 230 answer this question, we turn to a formal statistical detection framework, where we use the 231 unique ensemble mean "fingerprint" of the growth rate in the model sampled at flask sites 232 (i.e., the V-shaped dip and recovery in the ensemble-mean growth rate over January 2020 to 233 December 2021 in Figure 2) and quantify the correlation of each individual ensemble mem-234 ber with this fingerprint for each emissions scenario. This statistical detection approach for 235 hypothetical observations (we haven't yet measured the growth rate in December 2021, for 236 example) is identical to the one outlined in Lovenduski et al. [2021] and mimics the approach 237 for the detection of a climate change signal in real-world observations [Bindoff and Stott, 238 2013]. The resulting correlation coefficients are shown in the subplots of Figure 2, where 239 small circles show the set of 30 Pearson's correlation coefficients (r) with the ensemble mean 240 fingerprint across the 30 ensemble members, and large circles show the mean correlation 241 coefficients [calculated using a Fisher's z transform; see Lovenduski et al., 2021] for each 242 COVID-like emissions scenario. For the model sampled at Mauna Loa, the mean correlation 243 coefficient for the COVID-like ensemble is 0.4 with a wide range; stronger emissions reduc-244 tions increase the mean correlation coefficient and narrow the range (subplot in Figure 2a), 245

suggesting a higher probability of detecting the fingerprint from a single ensemble member
or hypothetical observational record under higher emissions reductions. Indeed, the range
of correlation coefficients is only statistically different from zero under the 2 × COVID-like
and 4 × COVID-like emission scenarios (subplot in Figure 2a), indicating that significant detection of the COVID fingerprint is only formally possible in cases with more extreme emissions reductions than those that occurred during the COVID pandemic. Similar patterns are
observed when the model is sampled at 12 flask sites (Figure 2b), though the correlations are
overall higher due to reduced internal variability.

Model-estimated, column-averaged dry-air mole fraction of χCO_2 [referred to as XCO₂ by the satellite community; Crisp et al., 2004] averaged over the extratropical Northern and 255 Southern Hemispheres (20°N-55°N and 20°S-55°S, respectively) shows only a small sig-256 nal of COVID-like emissions reductions amid large internal variability (Figure 3). While 257 the emissions reduction signal is more pronounced in the Southern Hemisphere extratrop-258 ics, there is large overlap of the various model ensemble members from the various emission 259 scenarios (Figure 3b), and only the $4 \times \text{COVID-like}$ ensemble mean is significantly different 260 from the control ensemble mean at the 2σ (95%) level (not shown). The vertical integration of the atmospheric column and the diffusive nature of atmospheric transport makes the mod-262 eled column concentrations less sensitive to changes in the surface emissions signal [Rayner 263 and O'Brien, 2001; Miller et al., 2007], thus making the signal more difficult to detect in the 264 column. 265

Carbon-concentration feedbacks further obscure the detection of COVID emissions 266 reductions in the atmospheric carbon reservoir. Figure 4 shows the anomaly in the ensemble-267 mean cumulative change in the modeled atmosphere, ocean, and land carbon reservoirs from 268 December 2019 to December 2040, where the anomaly is calculated relative to the control 269 ensemble mean. In the atmosphere, the cumulative reservoir anomaly is negative for the du-270 ration of the simulations regardless of emissions scenario (Figure 4a), indicating that each 271 of the COVID-like emissions perturbations leads to a forced change in the cumulative at-272 mospheric reservoir lasting well beyond the emissions recovery in 2022 (cf. Figure 4a and Figure S1b). For both the $2 \times \text{COVID-like}$ and $4 \times \text{COVID-like}$ scenarios, the atmosphere reservoir anomaly falls outside of the ensemble spread due to internal variability (gray shading) for several years. Meanwhile, the ocean and land carbon reservoirs also respond to the 276 COVID-like emissions reductions – the ocean carbon sink immediately slows with decreas-277 ing χCO_2 under all COVID-like scenarios [Figure 4b; Lovenduski et al., 2021], and the land 278 carbon sink also weakens, most noticeably under the $2 \times \text{COVID}$ -like and $4 \times \text{COVID}$ -like 279 scenarios (Figure 4c). The ocean reservoir anomaly falls outside of the spread due to internal 280 variability only in the 2 \times COVID-like and 4 \times COVID-like scenarios [Figure 4b; Lovenduski et al., 2021]. The land carbon sink anomaly is fully within the internal variability bounds 282 (Figure 4c), due to high internal variability in the land-air CO₂ flux [Denman et al., 2007]. 283 Nevertheless, these results suggest that COVID emissions reductions cause both the ocean 284 and land to absorb less carbon than usual in our model, thus reducing the perturbation in 285 the atmosphere. To illustrate this point further, we estimate the ensemble-mean, cumulative 286 change in the atmospheric carbon reservoir due only to emissions changes and plot the re-287 sulting reservoir anomaly as dashed lines in Figure 4a. This illustration reveals a critical role 288 for carbon-concentration feedbacks in the detection of COVID-driven emissions reductions: if not for the slowing ocean and land carbon sinks, the COVID-like emissions reduction sig-290 nal in the atmospheric carbon reservoir would have been detectable above the noise of inter-291 nal variability for three consecutive years (2022-2025), and for longer durations with larger 292 emission perturbations (Figure 4a). 293

4 Conclusions and Discussion

We use an initial-condition large ensemble of an Earth system model to assess the detectability of the COVID-driven emissions reductions signal in measurements of atmospheric CO₂ above the noise of internal variability and carbon-concentration feedbacks. We find a

unique fingerprint in atmospheric CO₂ growth rates calculated from simulated χ CO₂ mea-298 surements under COVID-like emissions reductions. The largest negative anomalies in the 299 atmospheric χCO_2 growth rate appear in February 2021, ~9 months after the peak emissions 300 reductions. This growth rate signal is more likely to be detected above the noise of internal variability when averaging over global flask network sites, rather than at an individual 302 site. However, this unique fingerprint is not formally detectable using a climate signal de-303 tection statistical approach, unless we force the model with unrealistically large emissions 304 reductions. Internal variability obscures the detection of change in the interhemispheric 305 difference of simulated χCO_2 from flask measurements and the simulated extra-tropical 306 column-average XCO₂ from satellite observations. Carbon-concentration feedbacks further 307 reduce the emissions signal in the atmospheric carbon reservoir. When we omit the effects of 308 these feedbacks on the atmospheric carbon reservoir, the signal in the cumulative reservoir 309 anomaly is detectable above the noise of internal variability over a three consecutive year 310 period (2022-2025). 311

Our study illuminates the challenges associated with detecting brief CO₂ emissions 312 reductions in global-scale atmospheric CO₂ from our established observational measure-313 ment systems. In order to see the emergence of the signal of COVID-driven emissions re-314 ductions in atmospheric CO_2 , one needs to first remove the influence of internal climate 315 variability and carbon-concentration feedbacks from the atmospheric CO₂ measurements. 316 While we are getting closer to quantifying the internal contribution to the total signal from 317 our measurements and producing near-real time estimates of this variability [e.g., Betts et al., 318 2016, 2020], we are not yet capable of quantifying carbon feedbacks from our current, ex-319 ploratory observational system [e.g., Sellers et al., 2018]. Further, the ocean and terrestrial 320 carbon reservoirs are only sparsely observed and, with the exception of a few surface ocean pCO₂ buoys [Sutton et al., 2019], the high-quality estimates of changing air-sea and air-land 322 CO2 fluxes that are available in a historical context are not yet available in near-real time 323 due to the high costs of fast data dissemination and other impediments. As we move into a 324 world characterized by intentional emissions reductions associated with international climate 325 change mitigation policies, we should consider this measurement infrastructure in the ocean 326 and terrestrial biosphere to detect the signal and monitor the impact of intentional emissions 327 reductions in atmospheric CO₂.

329 Acknowledgments

This research was supported by the National Science Foundation (OCE-1752724 and OCE-

³³¹ 1948664). A.C. was supported by funding from the NASA Grant/Cooperative Agreement

³³² 80NSSC20K0006. The data from the CanESM5 simulations used in this study are published

through the Government of Canada Open Data Portal, and can be accessed at http://crd-

data-donnees-rdc.ec.gc.ca/CCCMA/publications/COVID19/. We acknowledge the CCCma

staff who contributed to producing these simulations. The χ CO₂ flask data are from the

- Scripps CO₂ program, and can be accessed at *https://scrippsco2.ucsd.edu/data/atmospheric*
- ₃₃₇ _co2/sampling_stations.html.

338 References

Arora, V. K., G. J. Boer, P. Friedlingstein, M. Eby, C. D. Jones, J. R. Christian, G. Bonan,

L. Bopp, V. Brovkin, P. Cadule, T. Hajima, T. Ilyina, K. Lindsay, J. F. Tjiputra, and T. Wu

(2013), Carbon–Concentration and Carbon–Climate Feedbacks in CMIP5 Earth System
 Models, J. Climate, 26(15), 5289–5314, doi:10.1175/JCLI-D-12-00494.1.

³⁴² Models, J. Climate, 26(15), 5289–5314, doi:10.1175/JCLI-D-12-00494.1.

Arora, V. K., A. Katavouta, R. G. Williams, C. D. Jones, V. Brovkin, P. Friedlingstein,

J. Schwinger, L. Bopp, O. Boucher, P. Cadule, M. A. Chamberlain, J. R. Christian,

C. Delire, R. A. Fisher, T. Hajima, T. Ilyina, E. Joetzjer, M. Kawamiya, C. D. Koven, J. P.
 Krasting, R. M. Law, D. M. Lawrence, A. Lenton, K. Lindsay, J. Pongratz, T. Raddatz,

Krasting, R. M. Law, D. M. Lawrence, A. Lenton, K. Lindsay, J. Pongratz, T. Raddatz,

R. Séférian, K. Tachiiri, J. F. Tjiputra, A. Wiltshire, T. Wu, and T. Ziehn (2020), Carbon-

concentration and carbon–climate feedbacks in CMIP6 models and their comparison to

349	CMIP5 models, <i>Biogeosciences</i> , 17(16), 4173–4222, doi:10.5194/bg-17-4173-2020.
350	Betts, R., C. Jones, Y. Jin, R. Keeling, J. Kennedy, J. Knight, and A. Scaife (2020). Analy-
351	sis: What impact will the coronavirus pandemic have on atmospheric CO ₂ ? <i>CarbonBrief</i>
352	https://www.carbonbrief.org/analysis-what-impact-will-the-coronavirus-pandemic-have-
252	on-atmospheric-co?
355	Retts P. A. C. D. Jones, I. R. Knight, P. F. Keeling, and I. I. Kennedy (2016). El Niño and
354	a record CO., rise Nature Clim. Change 6, 806 ED
355	a record CO ₂ rise, <i>Nature Clini</i> . Change, 0, 800 Er –.
356	Change: from Clobal to Degional in Climate Change 2012: The Dhysical Science Pagis
357	Contribution of Working Group L to the Fifth Assessment Penort of the Intergovernmental
358	Contribution of working Group 1 to the Fifth Assessment Report of the Intergovernmental
359	Panel on Climate Change, edited by I. F. Slocker, D. Qill, GK. Plattier, M. M. B. Hg-
360	Combridge University Dress, Combridge, United Vingdom and New York, NY, USA
361	Cambridge University Fless, Cambridge, United Knigdolli and New Tork, NT, USA.
362	Buchwitz, M., M. Reuter, S. Noei, K. Bramstedt, O. Schneising, M. Hilker, B. Fuentes An-
363	drade, H. Bovensmann, J. P. Burrows, A. Di Noia, H. Boescn, L. wu, J. Landgraf, I. Aben,
364	C. Retscher, C. w. O Dell, and D. Crisp (2021), Can a regional-scale reduction of atmo-
365	spheric CO_2 during the COVID-19 pandemic be detected from space? A case study for
366	East Uning using salenne AUO_2 retrievals, Atmos. Meas. Iech., 14(3), 2141–2166, doi: 10.5104/omt.14.2141.2021
367	10.5194/amf-14-2141-2021.
368	Cadule, P., P. Friedlingstein, L. Bopp, S. Sitch, C. D. Jones, P. Ciais, S. L. Piao, and
369	P. Peylin (2010), Benchmarking coupled climate-carbon models against long-
370	term atmospheric CO_2 measurements, <i>Global Biogeochem</i> . Cycles, 24(2), doi:
371	https://doi.org/10.1029/2009GB003556.
372	Chatterjee, A., M. M. Gierach, A. J. Sutton, R. A. Feely, D. Crisp, A. Eldering, M. R. Gun-
373	son, C. W. O'Dell, B. B. Stephens, and D. S. Schimel (2017), Influence of El Niño on at-
374	mospheric CO_2 over the tropical Pacific Ocean: Findings from NASA's OCO-2 mission,
375	<i>Science</i> , 358(6360), eaam57/6, doi:10.1126/science.aam57/6.
376	Chevallier, F., B. Zheng, G. Broquet, P. Ciais, Z. Liu, S. J. Davis, Z. Deng, Y. Wang,
377	FM. Bréon, and C. W. O'Dell (2020), Local anomalies in the column-averaged
378	dry air mole fractions of carbon dioxide across the globe during the first months
379	of the Coronavirus Recession, <i>Geophys. Res. Lett.</i> , 47(22), e2020GL090,244, doi:
380	https://doi.org/10.1029/2020GL090244.
381	Ciais, P., J. Tan, X. Wang, C. Roedenbeck, F. Chevallier, S. L. Piao, R. Moriarty, G. Broquet,
381 382	Ciais, P., J. Tan, X. Wang, C. Roedenbeck, F. Chevallier, S. L. Piao, R. Moriarty, G. Broquet, C. Le Quéré, J. G. Canadell, S. Peng, B. Poulter, Z. Liu, and P. Tans (2019), Five decades
381 382 383	Ciais, P., J. Tan, X. Wang, C. Roedenbeck, F. Chevallier, S. L. Piao, R. Moriarty, G. Broquet, C. Le Quéré, J. G. Canadell, S. Peng, B. Poulter, Z. Liu, and P. Tans (2019), Five decades of northern land carbon uptake revealed by the interhemispheric CO ₂ gradient, <i>Nature</i> , 569(77751), 221, 225, p. 1010209(11596, 010, 10706).
381 382 383 384	Ciais, P., J. Tan, X. Wang, C. Roedenbeck, F. Chevallier, S. L. Piao, R. Moriarty, G. Broquet, C. Le Quéré, J. G. Canadell, S. Peng, B. Poulter, Z. Liu, and P. Tans (2019), Five decades of northern land carbon uptake revealed by the interhemispheric CO ₂ gradient, <i>Nature</i> , 568(7751), 221–225, doi:10.1038/s41586-019-1078-6.
381 382 383 384 385	 Ciais, P., J. Tan, X. Wang, C. Roedenbeck, F. Chevallier, S. L. Piao, R. Moriarty, G. Broquet, C. Le Quéré, J. G. Canadell, S. Peng, B. Poulter, Z. Liu, and P. Tans (2019), Five decades of northern land carbon uptake revealed by the interhemispheric CO₂ gradient, <i>Nature</i>, 568(7751), 221–225, doi:10.1038/s41586-019-1078-6. Crisp, D., R. Atlas, FM. Breon, L. Brown, J. Burrows, P. Ciais, B. Connor, S. Doney, D. B. L. D. M. B. Connor, S. Doney, D. B. L. D. M. B. Connor, S. Doney, D. B. L. D. M. B. Connor, S. D. B. Connor, S. Doney, D. B. L. B. Connor, S. D. B
381 382 383 384 385 386	 Ciais, P., J. Tan, X. Wang, C. Roedenbeck, F. Chevallier, S. L. Piao, R. Moriarty, G. Broquet, C. Le Quéré, J. G. Canadell, S. Peng, B. Poulter, Z. Liu, and P. Tans (2019), Five decades of northern land carbon uptake revealed by the interhemispheric CO₂ gradient, <i>Nature</i>, 568(7751), 221–225, doi:10.1038/s41586-019-1078-6. Crisp, D., R. Atlas, FM. Breon, L. Brown, J. Burrows, P. Ciais, B. Connor, S. Doney, I. Fung, D. Jacob, C. Miller, D. O'Brien, S. Pawson, J. Randerson, P. Rayner, R. Salawitz, S. Salawitz, S.
381 382 383 384 385 386 386	 Ciais, P., J. Tan, X. Wang, C. Roedenbeck, F. Chevallier, S. L. Piao, R. Moriarty, G. Broquet, C. Le Quéré, J. G. Canadell, S. Peng, B. Poulter, Z. Liu, and P. Tans (2019), Five decades of northern land carbon uptake revealed by the interhemispheric CO₂ gradient, <i>Nature</i>, 568(7751), 221–225, doi:10.1038/s41586-019-1078-6. Crisp, D., R. Atlas, FM. Breon, L. Brown, J. Burrows, P. Ciais, B. Connor, S. Doney, I. Fung, D. Jacob, C. Miller, D. O'Brien, S. Pawson, J. Randerson, P. Rayner, R. Salawitch, S. Sander, B. Sen, G. Stephens, P. Tans, G. Toon, P. Wennberg, S. Wofsy, Y. Yung, Z. Kurang, D. Chudenara, C. Sander, B. P. Wein, P. P. Wein, P. P. Wein, S. Y. Yung, C. Sander, C. Sander, R. Salawitch, S. Sander, B. Sen, G. Stephens, P. Tans, G. Done, P. Wein, S. Y. Yung, T. Kurang, D. Chudenara, C. Sander, P. P. Wein, P. P. Wein, P. P. Wein, S. Y. Yung, T. Kurang, P. Chudenara, C. Sander, P. P. Wein, P. P. Wein, S. Y. Yung, T. Kurang, P. Chudenara, C. Sander, P. P. Wein, P. P. Wein, S. Yung, Y. Yung, T. Kurang, P. Chudenara, C. Sander, P. P. Wein, P. P. Wein, P. P. Wein, S. Yung, T. Kurang, P. Chudenara, C. Sander, P. P. Wein, P. P. Wein, P. P. Wein, S. Yung, T. Kurang, P. Chudenara, C. Sander, P. P. Wein, P. P. Wein, P. P. Wein, P. Market, S. Sander, Sander,
381 382 383 384 385 386 387 388	 Ciais, P., J. Tan, X. Wang, C. Roedenbeck, F. Chevallier, S. L. Piao, R. Moriarty, G. Broquet, C. Le Quéré, J. G. Canadell, S. Peng, B. Poulter, Z. Liu, and P. Tans (2019), Five decades of northern land carbon uptake revealed by the interhemispheric CO₂ gradient, <i>Nature</i>, 568(7751), 221–225, doi:10.1038/s41586-019-1078-6. Crisp, D., R. Atlas, FM. Breon, L. Brown, J. Burrows, P. Ciais, B. Connor, S. Doney, I. Fung, D. Jacob, C. Miller, D. O'Brien, S. Pawson, J. Randerson, P. Rayner, R. Salawitch, S. Sander, B. Sen, G. Stephens, P. Tans, G. Toon, P. Wennberg, S. Wofsy, Y. Yung, Z. Kuang, B. Chudasama, G. Sprague, B. Weiss, R. Pollock, D. Kenyon, and S. Schroll
381 382 383 384 385 386 387 388 388	 Ciais, P., J. Tan, X. Wang, C. Roedenbeck, F. Chevallier, S. L. Piao, R. Moriarty, G. Broquet, C. Le Quéré, J. G. Canadell, S. Peng, B. Poulter, Z. Liu, and P. Tans (2019), Five decades of northern land carbon uptake revealed by the interhemispheric CO₂ gradient, <i>Nature</i>, 568(7751), 221–225, doi:10.1038/s41586-019-1078-6. Crisp, D., R. Atlas, FM. Breon, L. Brown, J. Burrows, P. Ciais, B. Connor, S. Doney, I. Fung, D. Jacob, C. Miller, D. O'Brien, S. Pawson, J. Randerson, P. Rayner, R. Salawitch, S. Sander, B. Sen, G. Stephens, P. Tans, G. Toon, P. Wennberg, S. Wofsy, Y. Yung, Z. Kuang, B. Chudasama, G. Sprague, B. Weiss, R. Pollock, D. Kenyon, and S. Schroll (2004), The orbiting carbon observatory (oco) mission, <i>Adv. Space Res.</i>, 34(4), 700–709, doi/httms//doi/org/10.1016/j.acg.2002.08.062
381 382 383 384 385 386 387 388 389 389	 Ciais, P., J. Tan, X. Wang, C. Roedenbeck, F. Chevallier, S. L. Piao, R. Moriarty, G. Broquet, C. Le Quéré, J. G. Canadell, S. Peng, B. Poulter, Z. Liu, and P. Tans (2019), Five decades of northern land carbon uptake revealed by the interhemispheric CO₂ gradient, <i>Nature</i>, 568(7751), 221–225, doi:10.1038/s41586-019-1078-6. Crisp, D., R. Atlas, FM. Breon, L. Brown, J. Burrows, P. Ciais, B. Connor, S. Doney, I. Fung, D. Jacob, C. Miller, D. O'Brien, S. Pawson, J. Randerson, P. Rayner, R. Salawitch, S. Sander, B. Sen, G. Stephens, P. Tans, G. Toon, P. Wennberg, S. Wofsy, Y. Yung, Z. Kuang, B. Chudasama, G. Sprague, B. Weiss, R. Pollock, D. Kenyon, and S. Schroll (2004), The orbiting carbon observatory (oco) mission, <i>Adv. Space Res.</i>, 34(4), 700–709, doi:https://doi.org/10.1016/j.asr.2003.08.062, trace Constituents in the Troposphere and Leven Set transment.
381 382 383 384 385 386 387 388 389 390 391	 Ciais, P., J. Tan, X. Wang, C. Roedenbeck, F. Chevallier, S. L. Piao, R. Moriarty, G. Broquet, C. Le Quéré, J. G. Canadell, S. Peng, B. Poulter, Z. Liu, and P. Tans (2019), Five decades of northern land carbon uptake revealed by the interhemispheric CO₂ gradient, <i>Nature</i>, 568(7751), 221–225, doi:10.1038/s41586-019-1078-6. Crisp, D., R. Atlas, FM. Breon, L. Brown, J. Burrows, P. Ciais, B. Connor, S. Doney, I. Fung, D. Jacob, C. Miller, D. O'Brien, S. Pawson, J. Randerson, P. Rayner, R. Salawitch, S. Sander, B. Sen, G. Stephens, P. Tans, G. Toon, P. Wennberg, S. Wofsy, Y. Yung, Z. Kuang, B. Chudasama, G. Sprague, B. Weiss, R. Pollock, D. Kenyon, and S. Schroll (2004), The orbiting carbon observatory (oco) mission, <i>Adv. Space Res.</i>, <i>34</i>(4), 700–709, doi:https://doi.org/10.1016/j.asr.2003.08.062, trace Constituents in the Troposphere and Lower Stratosphere.
381 382 383 384 385 386 387 388 389 390 391 392	 Ciais, P., J. Tan, X. Wang, C. Roedenbeck, F. Chevallier, S. L. Piao, R. Moriarty, G. Broquet, C. Le Quéré, J. G. Canadell, S. Peng, B. Poulter, Z. Liu, and P. Tans (2019), Five decades of northern land carbon uptake revealed by the interhemispheric CO₂ gradient, <i>Nature</i>, 568(7751), 221–225, doi:10.1038/s41586-019-1078-6. Crisp, D., R. Atlas, FM. Breon, L. Brown, J. Burrows, P. Ciais, B. Connor, S. Doney, I. Fung, D. Jacob, C. Miller, D. O'Brien, S. Pawson, J. Randerson, P. Rayner, R. Salawitch, S. Sander, B. Sen, G. Stephens, P. Tans, G. Toon, P. Wennberg, S. Wofsy, Y. Yung, Z. Kuang, B. Chudasama, G. Sprague, B. Weiss, R. Pollock, D. Kenyon, and S. Schroll (2004), The orbiting carbon observatory (oco) mission, <i>Adv. Space Res.</i>, <i>34</i>(4), 700–709, doi:https://doi.org/10.1016/j.asr.2003.08.062, trace Constituents in the Troposphere and Lower Stratosphere. Dargaville, R. J., S. C. Doney, and I. Y. Fung (2003), Inter-annual variability in the inter-
381 382 383 384 385 386 387 388 389 390 391 391 392 393	 Ciais, P., J. Tan, X. Wang, C. Roedenbeck, F. Chevallier, S. L. Piao, R. Moriarty, G. Broquet, C. Le Quéré, J. G. Canadell, S. Peng, B. Poulter, Z. Liu, and P. Tans (2019), Five decades of northern land carbon uptake revealed by the interhemispheric CO₂ gradient, <i>Nature</i>, 568(7751), 221–225, doi:10.1038/s41586-019-1078-6. Crisp, D., R. Atlas, FM. Breon, L. Brown, J. Burrows, P. Ciais, B. Connor, S. Doney, I. Fung, D. Jacob, C. Miller, D. O'Brien, S. Pawson, J. Randerson, P. Rayner, R. Salawitch, S. Sander, B. Sen, G. Stephens, P. Tans, G. Toon, P. Wennberg, S. Wofsy, Y. Yung, Z. Kuang, B. Chudasama, G. Sprague, B. Weiss, R. Pollock, D. Kenyon, and S. Schroll (2004), The orbiting carbon observatory (oco) mission, <i>Adv. Space Res.</i>, <i>34</i>(4), 700–709, doi:https://doi.org/10.1016/j.asr.2003.08.062, trace Constituents in the Troposphere and Lower Stratosphere. Dargaville, R. J., S. C. Doney, and I. Y. Fung (2003), Inter-annual variability in the interhemispheric atmospheric CO₂ gradient: contributions from transport and the seasonal recture.
381 382 383 384 385 386 387 388 389 390 391 392 393 394	 Ciais, P., J. Tan, X. Wang, C. Roedenbeck, F. Chevallier, S. L. Piao, R. Moriarty, G. Broquet, C. Le Quéré, J. G. Canadell, S. Peng, B. Poulter, Z. Liu, and P. Tans (2019), Five decades of northern land carbon uptake revealed by the interhemispheric CO₂ gradient, <i>Nature</i>, 568(7751), 221–225, doi:10.1038/s41586-019-1078-6. Crisp, D., R. Atlas, FM. Breon, L. Brown, J. Burrows, P. Ciais, B. Connor, S. Doney, I. Fung, D. Jacob, C. Miller, D. O'Brien, S. Pawson, J. Randerson, P. Rayner, R. Salawitch, S. Sander, B. Sen, G. Stephens, P. Tans, G. Toon, P. Wennberg, S. Wofsy, Y. Yung, Z. Kuang, B. Chudasama, G. Sprague, B. Weiss, R. Pollock, D. Kenyon, and S. Schroll (2004), The orbiting carbon observatory (oco) mission, <i>Adv. Space Res.</i>, <i>34</i>(4), 700–709, doi:https://doi.org/10.1016/j.asr.2003.08.062, trace Constituents in the Troposphere and Lower Stratosphere. Dargaville, R. J., S. C. Doney, and I. Y. Fung (2003), Inter-annual variability in the interhemispheric atmospheric CO₂ gradient: contributions from transport and the seasonal rectifier, <i>Tellus B Chem. Phys. Meteorol.</i>, <i>55</i>(2), 711–722, doi:10.3402/tellusb.v55i2.16713.
381 382 383 384 385 386 387 388 389 390 391 391 392 393 394 395	 Ciais, P., J. Tan, X. Wang, C. Roedenbeck, F. Chevallier, S. L. Piao, R. Moriarty, G. Broquet, C. Le Quéré, J. G. Canadell, S. Peng, B. Poulter, Z. Liu, and P. Tans (2019), Five decades of northern land carbon uptake revealed by the interhemispheric CO₂ gradient, <i>Nature</i>, 568(7751), 221–225, doi:10.1038/s41586-019-1078-6. Crisp, D., R. Atlas, FM. Breon, L. Brown, J. Burrows, P. Ciais, B. Connor, S. Doney, I. Fung, D. Jacob, C. Miller, D. O'Brien, S. Pawson, J. Randerson, P. Rayner, R. Salawitch, S. Sander, B. Sen, G. Stephens, P. Tans, G. Toon, P. Wennberg, S. Wofsy, Y. Yung, Z. Kuang, B. Chudasama, G. Sprague, B. Weiss, R. Pollock, D. Kenyon, and S. Schroll (2004), The orbiting carbon observatory (oco) mission, <i>Adv. Space Res.</i>, <i>34</i>(4), 700–709, doi:https://doi.org/10.1016/j.asr.2003.08.062, trace Constituents in the Troposphere and Lower Stratosphere. Dargaville, R. J., S. C. Doney, and I. Y. Fung (2003), Inter-annual variability in the interhemispheric atmospheric CO₂ gradient: contributions from transport and the seasonal rectifier, <i>Tellus B Chem. Phys. Meteorol.</i>, <i>55</i>(2), 711–722, doi:10.3402/tellusb.v55i2.16713. Denman, K. L., G. Brasseur, A. Chidthaisong, P. Ciais, P. M. Cox, R. E. Dickinson, P. M. Cox, P.
381 382 383 384 385 386 387 388 389 390 391 392 393 394 395 396	 Ciais, P., J. Tan, X. Wang, C. Roedenbeck, F. Chevallier, S. L. Piao, R. Moriarty, G. Broquet, C. Le Quéré, J. G. Canadell, S. Peng, B. Poulter, Z. Liu, and P. Tans (2019), Five decades of northern land carbon uptake revealed by the interhemispheric CO₂ gradient, <i>Nature</i>, 568(7751), 221–225, doi:10.1038/s41586-019-1078-6. Crisp, D., R. Atlas, FM. Breon, L. Brown, J. Burrows, P. Ciais, B. Connor, S. Doney, I. Fung, D. Jacob, C. Miller, D. O'Brien, S. Pawson, J. Randerson, P. Rayner, R. Salawitch, S. Sander, B. Sen, G. Stephens, P. Tans, G. Toon, P. Wennberg, S. Wofsy, Y. Yung, Z. Kuang, B. Chudasama, G. Sprague, B. Weiss, R. Pollock, D. Kenyon, and S. Schroll (2004), The orbiting carbon observatory (oco) mission, <i>Adv. Space Res.</i>, <i>34</i>(4), 700–709, doi:https://doi.org/10.1016/j.asr.2003.08.062, trace Constituents in the Troposphere and Lower Stratosphere. Dargaville, R. J., S. C. Doney, and I. Y. Fung (2003), Inter-annual variability in the interhemispheric atmospheric CO₂ gradient: contributions from transport and the seasonal rectifier, <i>Tellus B Chem. Phys. Meteorol.</i>, <i>55</i>(2), 711–722, doi:10.3402/tellusb.v55i2.16713. Denman, K. L., G. Brasseur, A. Chidthaisong, P. Ciais, P. M. Cox, R. E. Dickinson, D. Hauglustaine, C. Heinze, E. Holland, D. Jacob, U. Lohmann, S. Ramachandran, P. da
381 382 383 384 385 386 387 388 389 390 391 392 393 394 395 396 397	 Ciais, P., J. Tan, X. Wang, C. Roedenbeck, F. Chevallier, S. L. Piao, R. Moriarty, G. Broquet, C. Le Quéré, J. G. Canadell, S. Peng, B. Poulter, Z. Liu, and P. Tans (2019), Five decades of northern land carbon uptake revealed by the interhemispheric CO₂ gradient, <i>Nature</i>, 568(7751), 221–225, doi:10.1038/s41586-019-1078-6. Crisp, D., R. Atlas, FM. Breon, L. Brown, J. Burrows, P. Ciais, B. Connor, S. Doney, I. Fung, D. Jacob, C. Miller, D. O'Brien, S. Pawson, J. Randerson, P. Rayner, R. Salawitch, S. Sander, B. Sen, G. Stephens, P. Tans, G. Toon, P. Wennberg, S. Wofsy, Y. Yung, Z. Kuang, B. Chudasama, G. Sprague, B. Weiss, R. Pollock, D. Kenyon, and S. Schroll (2004), The orbiting carbon observatory (oco) mission, <i>Adv. Space Res.</i>, <i>34</i>(4), 700–709, doi:https://doi.org/10.1016/j.asr.2003.08.062, trace Constituents in the Troposphere and Lower Stratosphere. Dargaville, R. J., S. C. Doney, and I. Y. Fung (2003), Inter-annual variability in the interhemispheric atmospheric CO₂ gradient: contributions from transport and the seasonal rectifier, <i>Tellus B Chem. Phys. Meteorol.</i>, <i>55</i>(2), 711–722, doi:10.3402/tellusb.v55i2.16713. Denman, K. L., G. Brasseur, A. Chidthaisong, P. Ciais, P. M. Cox, R. E. Dickinson, D. Hauglustaine, C. Heinze, E. Holland, D. Jacob, U. Lohmann, S. Ramachandran, P. da Silva Dias, S. C. Wofsy, and X. Zhang (2007), Couplings Between Changes in the Climate Review of Key and S. Schroll Complexity of the seasonal rescience of the seasonal contribution for the seasonal rectifier at the seasonal rectifier for the form of the seasonal rectifier, <i>Tellus B Chem. Phys. Meteorol.</i>, <i>55</i>(2), 711–722, doi:10.3402/tellusb.v55i2.16713.
 381 382 383 384 385 386 387 388 389 390 391 392 393 394 395 396 397 398 	 Ciais, P., J. Tan, X. Wang, C. Roedenbeck, F. Chevallier, S. L. Piao, R. Moriarty, G. Broquet, C. Le Quéré, J. G. Canadell, S. Peng, B. Poulter, Z. Liu, and P. Tans (2019), Five decades of northern land carbon uptake revealed by the interhemispheric CO₂ gradient, <i>Nature</i>, <i>568</i>(7751), 221–225, doi:10.1038/s41586-019-1078-6. Crisp, D., R. Atlas, FM. Breon, L. Brown, J. Burrows, P. Ciais, B. Connor, S. Doney, I. Fung, D. Jacob, C. Miller, D. O'Brien, S. Pawson, J. Randerson, P. Rayner, R. Salawitch, S. Sander, B. Sen, G. Stephens, P. Tans, G. Toon, P. Wennberg, S. Wofsy, Y. Yung, Z. Kuang, B. Chudasama, G. Sprague, B. Weiss, R. Pollock, D. Kenyon, and S. Schroll (2004), The orbiting carbon observatory (oco) mission, <i>Adv. Space Res.</i>, <i>34</i>(4), 700–709, doi:https://doi.org/10.1016/j.asr.2003.08.062, trace Constituents in the Troposphere and Lower Stratosphere. Dargaville, R. J., S. C. Doney, and I. Y. Fung (2003), Inter-annual variability in the interhemispheric atmospheric CO₂ gradient: contributions from transport and the seasonal rectifier, <i>Tellus B Chem. Phys. Meteorol.</i>, <i>55</i>(2), 711–722, doi:10.3402/tellusb.v55i2.16713. Denman, K. L., G. Brasseur, A. Chidthaisong, P. Ciais, P. M. Cox, R. E. Dickinson, D. Hauglustaine, C. Heinze, E. Holland, D. Jacob, U. Lohmann, S. Ramachandran, P. da Silva Dias, S. C. Wofsy, and X. Zhang (2007), Couplings Between Changes in the Climate System and Biogeochemistry, in <i>Climate Change 2007: The Physical Science Basis. Con-</i>
 381 382 383 384 385 386 387 388 389 390 391 392 393 394 395 396 397 398 399 	 Ciais, P., J. Tan, X. Wang, C. Roedenbeck, F. Chevallier, S. L. Piao, R. Moriarty, G. Broquet, C. Le Quéré, J. G. Canadell, S. Peng, B. Poulter, Z. Liu, and P. Tans (2019), Five decades of northern land carbon uptake revealed by the interhemispheric CO₂ gradient, <i>Nature</i>, <i>568</i>(7751), 221–225, doi:10.1038/s41586-019-1078-6. Crisp, D., R. Atlas, FM. Breon, L. Brown, J. Burrows, P. Ciais, B. Connor, S. Doney, I. Fung, D. Jacob, C. Miller, D. O'Brien, S. Pawson, J. Randerson, P. Rayner, R. Salawitch, S. Sander, B. Sen, G. Stephens, P. Tans, G. Toon, P. Wennberg, S. Wofsy, Y. Yung, Z. Kuang, B. Chudasama, G. Sprague, B. Weiss, R. Pollock, D. Kenyon, and S. Schroll (2004), The orbiting carbon observatory (oco) mission, <i>Adv. Space Res.</i>, <i>34</i>(4), 700–709, doi:https://doi.org/10.1016/j.asr.2003.08.062, trace Constituents in the Troposphere and Lower Stratosphere. Dargaville, R. J., S. C. Doney, and I. Y. Fung (2003), Inter-annual variability in the interhemispheric atmospheric CO₂ gradient: contributions from transport and the seasonal rectifier, <i>Tellus B Chem. Phys. Meteorol.</i>, <i>55</i>(2), 711–722, doi:10.3402/tellusb.v55i2.16713. Denman, K. L., G. Brasseur, A. Chidthaisong, P. Ciais, P. M. Cox, R. E. Dickinson, D. Hauglustaine, C. Heinze, E. Holland, D. Jacob, U. Lohmann, S. Ramachandran, P. da Silva Dias, S. C. Wofsy, and X. Zhang (2007), Couplings Between Changes in the Climate System and Biogeochemistry, in <i>Climate Change 2007: The Physical Science Basis. Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Device of the Intergovernmental Device of the Intergovernmental Device of Working Group I to the Fourth Assessment Report of the Intergovernmental Device of Working Group I to the Fourth Assessment Report of the Intergovernmental Device of Working Group I to the Fourth Assessment Report of the Intergovernmental Device of Working Group I to the Fourth Assessment Report of the Intergovernmental Device of Working Group I to the Science</i>
 381 382 383 384 385 386 387 388 389 390 391 392 393 394 395 396 397 398 399 400 	 Ciais, P., J. Tan, X. Wang, C. Roedenbeck, F. Chevallier, S. L. Piao, R. Moriarty, G. Broquet, C. Le Quéré, J. G. Canadell, S. Peng, B. Poulter, Z. Liu, and P. Tans (2019), Five decades of northern land carbon uptake revealed by the interhemispheric CO₂ gradient, <i>Nature</i>, 568(7751), 221–225, doi:10.1038/s41586-019-1078-6. Crisp, D., R. Atlas, FM. Breon, L. Brown, J. Burrows, P. Ciais, B. Connor, S. Doney, I. Fung, D. Jacob, C. Miller, D. O'Brien, S. Pawson, J. Randerson, P. Rayner, R. Salawitch, S. Sander, B. Sen, G. Stephens, P. Tans, G. Toon, P. Wennberg, S. Wofsy, Y. Yung, Z. Kuang, B. Chudasama, G. Sprague, B. Weiss, R. Pollock, D. Kenyon, and S. Schroll (2004), The orbiting carbon observatory (oco) mission, <i>Adv. Space Res.</i>, <i>34</i>(4), 700–709, doi:https://doi.org/10.1016/j.asr.2003.08.062, trace Constituents in the Troposphere and Lower Stratosphere. Dargaville, R. J., S. C. Doney, and I. Y. Fung (2003), Inter-annual variability in the interhemispheric atmospheric CO₂ gradient: contributions from transport and the seasonal rectifier, <i>Tellus B Chem. Phys. Meteorol.</i>, <i>55</i>(2), 711–722, doi:10.3402/tellusb.v55i2.16713. Denman, K. L., G. Brasseur, A. Chidthaisong, P. Ciais, P. M. Cox, R. E. Dickinson, D. Hauglustaine, C. Heinze, E. Holland, D. Jacob, U. Lohmann, S. Ramachandran, P. da Silva Dias, S. C. Wofsy, and X. Zhang (2007), Couplings Between Changes in the Climate System and Biogeochemistry, in <i>Climate Change 2007: The Physical Science Basis. Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental <i>Panel on Climate Change</i>, edited by S. Solomon, D. Qin, M. Manning, Z. Chen, M. Mar-</i>
 381 382 383 384 385 386 387 388 389 390 391 392 393 394 395 396 397 398 399 400 401 	 Ciais, P., J. Tan, X. Wang, C. Roedenbeck, F. Chevallier, S. L. Piao, R. Moriarty, G. Broquet, C. Le Quéré, J. G. Canadell, S. Peng, B. Poulter, Z. Liu, and P. Tans (2019), Five decades of northern land carbon uptake revealed by the interhemispheric CO₂ gradient, <i>Nature</i>, <i>568</i>(7751), 221–225, doi:10.1038/s41586-019-1078-6. Crisp, D., R. Atlas, FM. Breon, L. Brown, J. Burrows, P. Ciais, B. Connor, S. Doney, I. Fung, D. Jacob, C. Miller, D. O'Brien, S. Pawson, J. Randerson, P. Rayner, R. Salawitch, S. Sander, B. Sen, G. Stephens, P. Tans, G. Toon, P. Wennberg, S. Wofsy, Y. Yung, Z. Kuang, B. Chudasama, G. Sprague, B. Weiss, R. Pollock, D. Kenyon, and S. Schroll (2004), The orbiting carbon observatory (oco) mission, <i>Adv. Space Res.</i>, <i>34</i>(4), 700–709, doi:https://doi.org/10.1016/j.asr.2003.08.062, trace Constituents in the Troposphere and Lower Stratosphere. Dargaville, R. J., S. C. Doney, and I. Y. Fung (2003), Inter-annual variability in the interhemispheric atmospheric CO₂ gradient: contributions from transport and the seasonal rectifier, <i>Tellus B Chem. Phys. Meteorol.</i>, <i>55</i>(2), 711–722, doi:10.3402/tellusb.v55i2.16713. Denman, K. L., G. Brasseur, A. Chidthaisong, P. Ciais, P. M. Cox, R. E. Dickinson, D. Hauglustaine, C. Heinze, E. Holland, D. Jacob, U. Lohmann, S. Ramachandran, P. da Silva Dias, S. C. Wofsy, and X. Zhang (2007), Couplings Between Changes in the Climate System and Biogeochemistry, in <i>Climate Change 2007: The Physical Science Basis. Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change</i>, edited by S. Solomon, D. Qin, M. Manning, Z. Chen, M. Marquis, K. B. Averyt, M. Tignor, and H. L. Miller, Cambridge University Press, Cambridge, M. S. Solomor, D. Qin, M. Manning, Z. Chen, M. Marquis, K. B. Averyt, M. Tignor, and H. L. Miller, Cambridge University Press, Cambridge, M. S. Marguis, K. B. Averyt, M. Tignor, and H. L. Miller, Cambridge University Press, Cambridge, M. S. Solomor, D. Qin, M. Mann

403 404	Deser, C., A. Phillips, V. Bourdette, and H. Teng (2012a), Uncertainty in climate change projections: the role of internal variability, <i>Clim. Dynam.</i> , <i>38</i> (3-4), 527–546, doi:
405	10.1007/s00382-010-0977-x.
406	Deser, C., R. Knutti, S. Solomon, and A. S. Phillips (2012b), Communication of the role of
407	natural variability in future North American climate, Nature Clim. Change, 2(11), 775-
408	779.
409	Deser, C., L. Terray, and A. S. Phillips (2016), Forced and internal components of winter air
410 411	temperature trends over North America during the past 50 years: Mechanisms and impli- cations, J. Climate, 6, 2237–2258, doi:10.1175/JCLI-D-15-0304.1.
412	Deser, C., F. Lehner, K. B. Rodgers, T. Ault, T. L. Delworth, P. N. DiNezio, A. Fiore,
413	C. Frankignoul, J. C. Fyfe, D. E. Horton, J. E. Kay, R. Knutti, N. S. Lovenduski,
414	J. Marotzke, K. A. McKinnon, S. Minobe, J. Randerson, J. A. Screen, I. R. Simpson, and
415	M. Ting (2020), Insights from Earth system model initial-condition large ensembles and
416	future prospects, Nature Clim. Change, 10(4), 277-286, doi:10.1038/s41558-020-0731-2.
417	Forster, P. M., H. I. Forster, M. J. Evans, M. J. Gidden, C. D. Jones, C. A. Keller, R. D.
418	Lamboll, C. L. Quéré, J. Rogelj, D. Rosen, CF. Schleussner, T. B. Richardson, C. J.
419	Smith, and S. T. Turnock (2020), Current and future global climate impacts resulting from
420	COVID-19, Nature Clim. Change, doi:10.1038/s41558-020-0883-0.
421	Friedlingstein, P., P. Cox, R. Betts, L. Bopp, W. von Bloh, V. Brovkin, P. Cadule, S. Doney,
422	M. Eby, I. Fung, G. Bala, J. John, C. Jones, F. Joos, T. Kato, M. Kawamiya, W. Knorr,
423	K. Lindsay, H. D. Matthews, T. Raddatz, P. Rayner, C. Reick, E. Roeckner, KG. Schnit-
424	zler, R. Schnur, K. Strassmann, A. J. Weaver, C. Yoshikawa, and N. Zeng (2006), Climate-
425	carbon cycle feedback analysis: Results from the C4MIP model intercomparison, J. Cli-
426	<i>mate</i> , 19(14), 3337–3353, doi:10.1175/JCLI3800.1.
427	Friedlingstein, P., M. O'Sullivan, M. W. Jones, R. M. Andrew, J. Hauck, A. Olsen, G. P.
428	Peters, W. Peters, J. Pongratz, S. Sitch, C. Le Quéré, J. G. Canadell, P. Ciais, R. B. Jack-
429	son, S. Alin, L. E. O. C. Aragão, A. Arneth, V. Arora, N. R. Bates, M. Becker, A. Benoit-
430	Cattin, H. C. Bittig, L. Bopp, S. Bultan, N. Chandra, F. Chevallier, L. P. Chini, W. Evans,
431	L. Florentie, P. M. Forster, I. Gasser, M. Gehlen, D. Gilfillan, I. Gkritzalis, L. Gre-
432	gor, N. Gruber, I. Harris, K. Harlung, V. Haverd, K. A. Houghton, I. Hyma, A. K. Jain,
433	E. Joetzjei, K. Kadollo, E. Kato, V. Kilidis, J. I. Kolsoakkell, F. Landschulzel, N. Lelevie,
434	I F M S Nabel S I Nakaoka Y Niwa K O'Brien T Ono P I Palmer D Pierrot
435	B Poulter I. Resplandy F. Robertson C. Rödenbeck I Schwinger R. Séférian I. Skiel-
430	van A I P Smith A I Sutton T Tanhua P P Tans H Tian B Tilbrook G van der
438	Werf, N. Vuichard, A. P. Walker, R. Wanninkhof, A. J. Watson, D. Willis, A. J. Wilt-
439	shire, W. Yuan, X. Yue, and S. Zaehle (2020), Global Carbon Budget 2020, <i>Earth Syst.</i>
440	Sci. Data, 12(4), 3269–3340, doi:10.5194/essd-12-3269-2020.
441	Frölicher, T. L., F. Joos, C. C. Raible, and J. L. Sarmiento (2013), Atmospheric CO ₂ re-
442	sponse to volcanic eruptions: The role of ENSO, season, and variability, Global Bio-
443	geochem. Cycles, 27(1), 239-251, doi:https://doi.org/10.1002/gbc.20028.
444	Fyfe, J. C., V. V. Kharin, N. Swart, G. M. Flato, M. Sigmond, and N. P. Gillett (2021), Quan-
445	tifying the influence of short-term emission reductions on climate, Science Advances,
446	7(10), eabf7133, doi:10.1126/sciadv.abf7133.
447	Gettelman, A., R. Lamboll, C. G. Bardeen, P. M. Forster, and D. Watson-Parris (2021),
448	Climate impacts of COVID-19 induced emission changes, Geophys. Res. Lett., 48(3),
449	e2020GL091,805, doi:https://doi.org/10.1029/2020GL091805.
450	Hawkins, E., and R. Sutton (2009), The potential to narrow uncertainty in regional climate
451	predictions, B. Am. Meteorol. Soc., 90(8), 1095-1107, doi:10.1175/2009BAMS2607.1.
452	Hoesly, R. M., S. J. Smith, L. Feng, Z. Klimont, G. Janssens-Maenhout, T. Pitkanen, J. J.
453	Seibert, L. Vu, R. J. Andres, R. M. Bolt, T. C. Bond, L. Dawidowski, N. Kholod, JI.
454	Kurokawa, M. Li, L. Liu, Z. Lu, M. C. P. Moura, P. R. O'Rourke, and Q. Zhang (2018),
455	Historical (1750-2014) anthropogenic emissions of reactive gases and aerosols from the

456 Community Emissions Data System (CEDS), *Geosci. Model Dev.*, 11(1), 369–408, doi:

457	10.5194/gmd-11-369-2018.
458	Ilyina, T., H. Li, A. Spring, W. A. Müller, L. Bopp, M. O. Chikamoto, G. Danabasoglu,
459	M. Dobrynin, J. Dunne, F. Fransner, P. Friedlingstein, W. Lee, N. S. Lovenduski, W. J.
460	Merryfield, J. Mignot, J. Y. Park, R. Séférian, R. Sospedra-Alfonso, M. Watanabe,
461	and S. Yeager (2021), Predictable variations of the carbon sinks and atmospheric CO_2
462	growth in a multi-model framework, Geophys. Res. Lett., 48(6), e2020GL090,695, doi:
463	https://doi.org/10.1029/2020GL090695.
464	International Energy Agency (2021), World Energy Investment 2021, Tech. rep., Paris,
465	https://www.iea.org/reports/world-energy-investment-2021.
466	Jones, C. D., J. E. Hickman, S. T. Rumbold, J. Walton, R. D. Lamboll, R. B. Skeie,
467	S. Fiedler, P. M. Forster, J. Rogelj, M. Abe, M. Botzet, K. Calvin, C. Cassou, J. N. S. Cole,
468	P. Davini, M. Deushi, M. Dix, J. C. Fyfe, N. P. Gillett, T. Ilyina, M. Kawamiya, M. Kel-
469	ley, S. Kharin, T. Koshiro, H. Li, C. Mackallah, W. A. Müller, P. Nabat, T. van Noije,
470	P. Nolan, R. Ohgaito, D. Olivié, N. Oshima, J. Parodi, T. J. Reerink, L. Ren, A. Romanou,
471	R. Séférian, Y. Tang, C. Timmreck, J. Tjiputra, E. Tourigny, K. Tsigaridis, H. Wang,
472	M. Wu, K. Wyser, S. Yang, Y. Yang, and T. Ziehn (2021), The climate response to emis-
473	sions reductions due to COVID-19: Initial results from CovidMIP, Geophys. Res. Lett.,
474	48(8), e2020GL091,883, doi:https://doi.org/10.1029/2020GL091883.
475	Keeling, C., S. Piper, R. Bacastow, M. Wahlen, T. Whorf, M. Heimann, and H. Meijer
476	(2001), Exchanges of atmospheric CO_2 and ${}^{13}CO_2$ with the terrestrial biosphere and
477	oceans from 1978 to 2000. i. global aspects, in UC San Diego: Library - Scripps Digital
478	Collection.
479	Le Quéré, C., R. B. Jackson, M. W. Jones, A. J. P. Smith, S. Abernethy, R. M. Andrew, A. J.
480	De-Gol, D. R. Willis, Y. Shan, J. G. Canadell, P. Friedlingstein, F. Creutzig, and G. P. Pe-
481	ters (2020), Temporary reduction in daily global CO ₂ emissions during the COVID-19 for a local base for emission of the COVID-19 for a local base for emission of the covid base for the covid base of the covi
482	Torced confinement, <i>Nature Clim. Change</i> , 10(7), 647–653, doi:10.1038/s41558-020-0797-
483	
400	A. La Quárá C. C. D. Datara D. Eriadlingatain, D. M. Andrew, J. C. Canadall, S. J. Davia, D. P.
484	Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson and M. W. Jones (2021). Fossil CO ₂ emissions in the post COVID 19 erg. <i>Nature</i>
484	Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson, and M. W. Jones (2021), Fossil CO ₂ emissions in the post-COVID-19 era, <i>Nature Clim Change</i> 11(3), 197–199. doi:10.1038/s41558-021-01001-0
484 485 486	 Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson, and M. W. Jones (2021), Fossil CO₂ emissions in the post-COVID-19 era, <i>Nature Clim. Change</i>, <i>11</i>(3), 197–199, doi:10.1038/s41558-021-01001-0. Liu D. W. Sun N. Zeng, P. Han, B. Yao, Z. Liu, P. Wang, K. Zheng, H. Mei, and O. Cai
484 485 486 487	 Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson, and M. W. Jones (2021), Fossil CO₂ emissions in the post-COVID-19 era, <i>Nature Clim. Change</i>, <i>11</i>(3), 197–199, doi:10.1038/s41558-021-01001-0. Liu, D., W. Sun, N. Zeng, P. Han, B. Yao, Z. Liu, P. Wang, K. Zheng, H. Mei, and Q. Cai (2021). Observed decreases in on-road CO₂ concentrations in Beijing during COVID-19
484 485 486 487 488 489	 Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson, and M. W. Jones (2021), Fossil CO₂ emissions in the post-COVID-19 era, <i>Nature Clim. Change</i>, <i>11</i>(3), 197–199, doi:10.1038/s41558-021-01001-0. Liu, D., W. Sun, N. Zeng, P. Han, B. Yao, Z. Liu, P. Wang, K. Zheng, H. Mei, and Q. Cai (2021), Observed decreases in on-road CO₂ concentrations in Beijing during COVID-19 restrictions. <i>Atmos. Chem. Phys.</i> 21(6), 4599–4614. doi:10.5194/acp-21-4599-2021
484 485 486 487 488 489	 Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson, and M. W. Jones (2021), Fossil CO₂ emissions in the post-COVID-19 era, <i>Nature Clim. Change</i>, <i>11</i>(3), 197–199, doi:10.1038/s41558-021-01001-0. Liu, D., W. Sun, N. Zeng, P. Han, B. Yao, Z. Liu, P. Wang, K. Zheng, H. Mei, and Q. Cai (2021), Observed decreases in on-road CO₂ concentrations in Beijing during COVID-19 restrictions, <i>Atmos. Chem. Phys.</i>, <i>21</i>(6), 4599–4614, doi:10.5194/acp-21-4599-2021. Liu, L. K. W. Bowman, D. S. Schimel, N. C. Parazoo, Z. Jiang, M. Lee, A. A. Bloom.
484 485 486 487 488 489 490 491	 Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson, and M. W. Jones (2021), Fossil CO₂ emissions in the post-COVID-19 era, <i>Nature Clim. Change</i>, <i>11</i>(3), 197–199, doi:10.1038/s41558-021-01001-0. Liu, D., W. Sun, N. Zeng, P. Han, B. Yao, Z. Liu, P. Wang, K. Zheng, H. Mei, and Q. Cai (2021), Observed decreases in on-road CO₂ concentrations in Beijing during COVID-19 restrictions, <i>Atmos. Chem. Phys.</i>, <i>21</i>(6), 4599–4614, doi:10.5194/acp-21-4599-2021. Liu, J., K. W. Bowman, D. S. Schimel, N. C. Parazoo, Z. Jiang, M. Lee, A. A. Bloom, D. Wunch, C. Frankenberg, Y. Sun, C. W. O'Dell, K. R. Gurney, D. Menemenlis.
484 485 486 487 488 489 490 491 492	 Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson, and M. W. Jones (2021), Fossil CO₂ emissions in the post-COVID-19 era, <i>Nature Clim. Change</i>, <i>11</i>(3), 197–199, doi:10.1038/s41558-021-01001-0. Liu, D., W. Sun, N. Zeng, P. Han, B. Yao, Z. Liu, P. Wang, K. Zheng, H. Mei, and Q. Cai (2021), Observed decreases in on-road CO₂ concentrations in Beijing during COVID-19 restrictions, <i>Atmos. Chem. Phys.</i>, <i>21</i>(6), 4599–4614, doi:10.5194/acp-21-4599-2021. Liu, J., K. W. Bowman, D. S. Schimel, N. C. Parazoo, Z. Jiang, M. Lee, A. A. Bloom, D. Wunch, C. Frankenberg, Y. Sun, C. W. O'Dell, K. R. Gurney, D. Menemenlis, M. Gierach, D. Crisp. and A. Eldering (2017). Contrasting carbon cycle responses of
484 485 486 487 488 489 490 491 491 492 493	 Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson, and M. W. Jones (2021), Fossil CO₂ emissions in the post-COVID-19 era, <i>Nature Clim. Change</i>, <i>11</i>(3), 197–199, doi:10.1038/s41558-021-01001-0. Liu, D., W. Sun, N. Zeng, P. Han, B. Yao, Z. Liu, P. Wang, K. Zheng, H. Mei, and Q. Cai (2021), Observed decreases in on-road CO₂ concentrations in Beijing during COVID-19 restrictions, <i>Atmos. Chem. Phys.</i>, <i>21</i>(6), 4599–4614, doi:10.5194/acp-21-4599-2021. Liu, J., K. W. Bowman, D. S. Schimel, N. C. Parazoo, Z. Jiang, M. Lee, A. A. Bloom, D. Wunch, C. Frankenberg, Y. Sun, C. W. O'Dell, K. R. Gurney, D. Menemenlis, M. Gierach, D. Crisp, and A. Eldering (2017), Contrasting carbon cycle responses of the tropical continents to the 2015–2016 El Niño, <i>Science</i>, <i>358</i>(6360), eaam5690, doi:
484 485 486 487 488 489 490 491 492 493 494	 Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson, and M. W. Jones (2021), Fossil CO₂ emissions in the post-COVID-19 era, <i>Nature Clim. Change</i>, <i>11</i>(3), 197–199, doi:10.1038/s41558-021-01001-0. Liu, D., W. Sun, N. Zeng, P. Han, B. Yao, Z. Liu, P. Wang, K. Zheng, H. Mei, and Q. Cai (2021), Observed decreases in on-road CO₂ concentrations in Beijing during COVID-19 restrictions, <i>Atmos. Chem. Phys.</i>, <i>21</i>(6), 4599–4614, doi:10.5194/acp-21-4599-2021. Liu, J., K. W. Bowman, D. S. Schimel, N. C. Parazoo, Z. Jiang, M. Lee, A. A. Bloom, D. Wunch, C. Frankenberg, Y. Sun, C. W. O'Dell, K. R. Gurney, D. Menemenlis, M. Gierach, D. Crisp, and A. Eldering (2017), Contrasting carbon cycle responses of the tropical continents to the 2015–2016 El Niño, <i>Science</i>, <i>358</i>(6360), eaam5690, doi: 10.1126/science.aam5690.
484 485 486 487 488 489 490 491 492 493 494 495	 Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson, and M. W. Jones (2021), Fossil CO₂ emissions in the post-COVID-19 era, <i>Nature Clim. Change</i>, <i>11</i>(3), 197–199, doi:10.1038/s41558-021-01001-0. Liu, D., W. Sun, N. Zeng, P. Han, B. Yao, Z. Liu, P. Wang, K. Zheng, H. Mei, and Q. Cai (2021), Observed decreases in on-road CO₂ concentrations in Beijing during COVID-19 restrictions, <i>Atmos. Chem. Phys.</i>, <i>21</i>(6), 4599–4614, doi:10.5194/acp-21-4599-2021. Liu, J., K. W. Bowman, D. S. Schimel, N. C. Parazoo, Z. Jiang, M. Lee, A. A. Bloom, D. Wunch, C. Frankenberg, Y. Sun, C. W. O'Dell, K. R. Gurney, D. Menemenlis, M. Gierach, D. Crisp, and A. Eldering (2017), Contrasting carbon cycle responses of the tropical continents to the 2015–2016 El Niño, <i>Science</i>, <i>358</i>(6360), eaam5690, doi: 10.1126/science.aam5690. Liu, Z., P. Ciais, Z. Deng, R. Lei, S. J. Davis, S. Feng, B. Zheng, D. Cui, X. Dou, B. Zhu,
484 485 486 487 488 489 490 491 492 493 494 495 496	 Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson, and M. W. Jones (2021), Fossil CO₂ emissions in the post-COVID-19 era, <i>Nature Clim. Change</i>, <i>11</i>(3), 197–199, doi:10.1038/s41558-021-01001-0. Liu, D., W. Sun, N. Zeng, P. Han, B. Yao, Z. Liu, P. Wang, K. Zheng, H. Mei, and Q. Cai (2021), Observed decreases in on-road CO₂ concentrations in Beijing during COVID-19 restrictions, <i>Atmos. Chem. Phys.</i>, <i>21</i>(6), 4599–4614, doi:10.5194/acp-21-4599-2021. Liu, J., K. W. Bowman, D. S. Schimel, N. C. Parazoo, Z. Jiang, M. Lee, A. A. Bloom, D. Wunch, C. Frankenberg, Y. Sun, C. W. O'Dell, K. R. Gurney, D. Menemenlis, M. Gierach, D. Crisp, and A. Eldering (2017), Contrasting carbon cycle responses of the tropical continents to the 2015–2016 El Niño, <i>Science</i>, <i>358</i>(6360), eaam5690, doi: 10.1126/science.aam5690. Liu, Z., P. Ciais, Z. Deng, R. Lei, S. J. Davis, S. Feng, B. Zheng, D. Cui, X. Dou, B. Zhu, R. Guo, P. Ke, T. Sun, C. Lu, P. He, Y. Wang, X. Yue, Y. Wang, Y. Lei, H. Zhou, Z. Cai,
484 485 486 487 488 489 490 491 492 493 493 494 495 496 497	 Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson, and M. W. Jones (2021), Fossil CO₂ emissions in the post-COVID-19 era, <i>Nature Clim. Change</i>, <i>11</i>(3), 197–199, doi:10.1038/s41558-021-01001-0. Liu, D., W. Sun, N. Zeng, P. Han, B. Yao, Z. Liu, P. Wang, K. Zheng, H. Mei, and Q. Cai (2021), Observed decreases in on-road CO₂ concentrations in Beijing during COVID-19 restrictions, <i>Atmos. Chem. Phys.</i>, <i>21</i>(6), 4599–4614, doi:10.5194/acp-21-4599-2021. Liu, J., K. W. Bowman, D. S. Schimel, N. C. Parazoo, Z. Jiang, M. Lee, A. A. Bloom, D. Wunch, C. Frankenberg, Y. Sun, C. W. O'Dell, K. R. Gurney, D. Menemenlis, M. Gierach, D. Crisp, and A. Eldering (2017), Contrasting carbon cycle responses of the tropical continents to the 2015–2016 El Niño, <i>Science</i>, <i>358</i>(6360), eaam5690, doi: 10.1126/science.aam5690. Liu, Z., P. Ciais, Z. Deng, R. Lei, S. J. Davis, S. Feng, B. Zheng, D. Cui, X. Dou, B. Zhu, R. Guo, P. Ke, T. Sun, C. Lu, P. He, Y. Wang, X. Yue, Y. Wang, Y. Lei, H. Zhou, Z. Cai, Y. Wu, R. Guo, T. Han, J. Xue, O. Boucher, E. Boucher, F. Chevallier, K. Tanaka, Y. Wei,
484 485 486 487 488 489 490 491 492 493 494 495 496 497 498	 Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson, and M. W. Jones (2021), Fossil CO₂ emissions in the post-COVID-19 era, <i>Nature Clim. Change</i>, <i>11</i>(3), 197–199, doi:10.1038/s41558-021-01001-0. Liu, D., W. Sun, N. Zeng, P. Han, B. Yao, Z. Liu, P. Wang, K. Zheng, H. Mei, and Q. Cai (2021), Observed decreases in on-road CO₂ concentrations in Beijing during COVID-19 restrictions, <i>Atmos. Chem. Phys.</i>, <i>21</i>(6), 4599–4614, doi:10.5194/acp-21-4599-2021. Liu, J., K. W. Bowman, D. S. Schimel, N. C. Parazoo, Z. Jiang, M. Lee, A. A. Bloom, D. Wunch, C. Frankenberg, Y. Sun, C. W. O'Dell, K. R. Gurney, D. Menemenlis, M. Gierach, D. Crisp, and A. Eldering (2017), Contrasting carbon cycle responses of the tropical continents to the 2015–2016 El Niño, <i>Science</i>, <i>358</i>(6360), eaam5690, doi: 10.1126/science.aam5690. Liu, Z., P. Ciais, Z. Deng, R. Lei, S. J. Davis, S. Feng, B. Zheng, D. Cui, X. Dou, B. Zhu, R. Guo, P. Ke, T. Sun, C. Lu, P. He, Y. Wang, X. Yue, Y. Wang, Y. Lei, H. Zhou, Z. Cai, Y. Wu, R. Guo, T. Han, J. Xue, O. Boucher, E. Boucher, F. Chevallier, K. Tanaka, Y. Wei, H. Zhong, C. Kang, N. Zhang, B. Chen, F. Xi, M. Liu, FM. Bréon, Y. Lu, Q. Zhang,
484 485 486 487 488 489 490 491 492 493 494 495 495 496 497 498 499	 Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson, and M. W. Jones (2021), Fossil CO₂ emissions in the post-COVID-19 era, <i>Nature Clim. Change</i>, <i>11</i>(3), 197–199, doi:10.1038/s41558-021-01001-0. Liu, D., W. Sun, N. Zeng, P. Han, B. Yao, Z. Liu, P. Wang, K. Zheng, H. Mei, and Q. Cai (2021), Observed decreases in on-road CO₂ concentrations in Beijing during COVID-19 restrictions, <i>Atmos. Chem. Phys.</i>, <i>21</i>(6), 4599–4614, doi:10.5194/acp-21-4599-2021. Liu, J., K. W. Bowman, D. S. Schimel, N. C. Parazoo, Z. Jiang, M. Lee, A. A. Bloom, D. Wunch, C. Frankenberg, Y. Sun, C. W. O'Dell, K. R. Gurney, D. Menemenlis, M. Gierach, D. Crisp, and A. Eldering (2017), Contrasting carbon cycle responses of the tropical continents to the 2015–2016 El Niño, <i>Science</i>, <i>358</i>(6360), eaam5690, doi: 10.1126/science.aam5690. Liu, Z., P. Ciais, Z. Deng, R. Lei, S. J. Davis, S. Feng, B. Zheng, D. Cui, X. Dou, B. Zhu, R. Guo, P. Ke, T. Sun, C. Lu, P. He, Y. Wang, X. Yue, Y. Wang, Y. Lei, H. Zhou, Z. Cai, Y. Wu, R. Guo, T. Han, J. Xue, O. Boucher, E. Boucher, F. Chevallier, K. Tanaka, Y. Wei, H. Zhong, C. Kang, N. Zhang, B. Chen, F. Xi, M. Liu, FM. Bréon, Y. Lu, Q. Zhang, D. Guan, P. Gong, D. M. Kammen, K. He, and H. J. Schellnhuber (2020), Near-real-time
484 485 486 487 488 489 490 491 492 493 494 495 495 496 497 498 499 500	 Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson, and M. W. Jones (2021), Fossil CO₂ emissions in the post-COVID-19 era, <i>Nature Clim. Change</i>, <i>11</i>(3), 197–199, doi:10.1038/s41558-021-01001-0. Liu, D., W. Sun, N. Zeng, P. Han, B. Yao, Z. Liu, P. Wang, K. Zheng, H. Mei, and Q. Cai (2021), Observed decreases in on-road CO₂ concentrations in Beijing during COVID-19 restrictions, <i>Atmos. Chem. Phys.</i>, <i>21</i>(6), 4599–4614, doi:10.5194/acp-21-4599-2021. Liu, J., K. W. Bowman, D. S. Schimel, N. C. Parazoo, Z. Jiang, M. Lee, A. A. Bloom, D. Wunch, C. Frankenberg, Y. Sun, C. W. O'Dell, K. R. Gurney, D. Menemenlis, M. Gierach, D. Crisp, and A. Eldering (2017), Contrasting carbon cycle responses of the tropical continents to the 2015–2016 El Niño, <i>Science</i>, <i>358</i>(6360), eaam5690, doi: 10.1126/science.aam5690. Liu, Z., P. Ciais, Z. Deng, R. Lei, S. J. Davis, S. Feng, B. Zheng, D. Cui, X. Dou, B. Zhu, R. Guo, P. Ke, T. Sun, C. Lu, P. He, Y. Wang, X. Yue, Y. Wang, Y. Lei, H. Zhou, Z. Cai, Y. Wu, R. Guo, T. Han, J. Xue, O. Boucher, E. Boucher, F. Chevallier, K. Tanaka, Y. Wei, H. Zhong, C. Kang, N. Zhang, B. Chen, F. Xi, M. Liu, FM. Bréon, Y. Lu, Q. Zhang, D. Guan, P. Gong, D. M. Kammen, K. He, and H. J. Schellnhuber (2020), Near-real-time monitoring of global CO₂ emissions reveals the effects of the COVID-19 pandemic, <i>Na</i>-
484 485 486 487 488 489 490 491 492 493 494 493 493 494 495 496 499 500 501	 Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson, and M. W. Jones (2021), Fossil CO₂ emissions in the post-COVID-19 era, <i>Nature Clim. Change</i>, <i>11</i>(3), 197–199, doi:10.1038/s41558-021-01001-0. Liu, D., W. Sun, N. Zeng, P. Han, B. Yao, Z. Liu, P. Wang, K. Zheng, H. Mei, and Q. Cai (2021), Observed decreases in on-road CO₂ concentrations in Beijing during COVID-19 restrictions, <i>Atmos. Chem. Phys.</i>, <i>21</i>(6), 4599–4614, doi:10.5194/acp-21-4599-2021. Liu, J., K. W. Bowman, D. S. Schimel, N. C. Parazoo, Z. Jiang, M. Lee, A. A. Bloom, D. Wunch, C. Frankenberg, Y. Sun, C. W. O'Dell, K. R. Gurney, D. Menemenlis, M. Gierach, D. Crisp, and A. Eldering (2017), Contrasting carbon cycle responses of the tropical continents to the 2015–2016 El Niño, <i>Science</i>, <i>358</i>(6360), eaam5690, doi: 10.1126/science.aam5690. Liu, Z., P. Ciais, Z. Deng, R. Lei, S. J. Davis, S. Feng, B. Zheng, D. Cui, X. Dou, B. Zhu, R. Guo, P. Ke, T. Sun, C. Lu, P. He, Y. Wang, X. Yue, Y. Wang, Y. Lei, H. Zhou, Z. Cai, Y. Wu, R. Guo, T. Han, J. Xue, O. Boucher, E. Boucher, F. Chevallier, K. Tanaka, Y. Wei, H. Zhong, C. Kang, N. Zhang, B. Chen, F. Xi, M. Liu, FM. Bréon, Y. Lu, Q. Zhang, D. Guan, P. Gong, D. M. Kammen, K. He, and H. J. Schellnhuber (2020), Near-real-time monitoring of global CO₂ emissions reveals the effects of the COVID-19 pandemic, <i>Nature Comm.</i>, <i>11</i>(1), 5172, doi:10.1038/s41467-020-18922-7.
484 485 486 487 488 489 490 491 492 493 493 493 494 495 496 499 500 501 502	 ^{A.} Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson, and M. W. Jones (2021), Fossil CO₂ emissions in the post-COVID-19 era, <i>Nature Clim. Change</i>, <i>11</i>(3), 197–199, doi:10.1038/s41558-021-01001-0. Liu, D., W. Sun, N. Zeng, P. Han, B. Yao, Z. Liu, P. Wang, K. Zheng, H. Mei, and Q. Cai (2021), Observed decreases in on-road CO₂ concentrations in Beijing during COVID-19 restrictions, <i>Atmos. Chem. Phys.</i>, <i>21</i>(6), 4599–4614, doi:10.5194/acp-21-4599-2021. Liu, J., K. W. Bowman, D. S. Schimel, N. C. Parazoo, Z. Jiang, M. Lee, A. A. Bloom, D. Wunch, C. Frankenberg, Y. Sun, C. W. O'Dell, K. R. Gurney, D. Menemenlis, M. Gierach, D. Crisp, and A. Eldering (2017), Contrasting carbon cycle responses of the tropical continents to the 2015–2016 El Niño, <i>Science</i>, <i>358</i>(6360), eaam5690, doi: 10.1126/science.aam5690. Liu, Z., P. Ciais, Z. Deng, R. Lei, S. J. Davis, S. Feng, B. Zheng, D. Cui, X. Dou, B. Zhu, R. Guo, P. Ke, T. Sun, C. Lu, P. He, Y. Wang, X. Yue, Y. Wang, Y. Lei, H. Zhou, Z. Cai, Y. Wu, R. Guo, T. Han, J. Xue, O. Boucher, E. Boucher, F. Chevallier, K. Tanaka, Y. Wei, H. Zhong, C. Kang, N. Zhang, B. Chen, F. Xi, M. Liu, FM. Bréon, Y. Lu, Q. Zhang, D. Guan, P. Gong, D. M. Kammen, K. He, and H. J. Schellnhuber (2020), Near-real-time monitoring of global CO₂ emissions reveals the effects of the COVID-19 pandemic, <i>Nature Comm.</i>, <i>11</i>(1), 5172, doi:10.1038/s41467-020-18922-7. Lovenduski, N. S., N. C. Swart, A. J. Sutton, J. C. Fyfe, G. A. McKinley, C. Sabine,
484 485 486 487 488 489 490 491 492 493 494 495 494 495 496 497 498 499 500 501 502 503	 ^{A.} Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson, and M. W. Jones (2021), Fossil CO₂ emissions in the post-COVID-19 era, <i>Nature Clim. Change</i>, <i>11</i>(3), 197–199, doi:10.1038/s41558-021-01001-0. Liu, D., W. Sun, N. Zeng, P. Han, B. Yao, Z. Liu, P. Wang, K. Zheng, H. Mei, and Q. Cai (2021), Observed decreases in on-road CO₂ concentrations in Beijing during COVID-19 restrictions, <i>Atmos. Chem. Phys.</i>, <i>21</i>(6), 4599–4614, doi:10.5194/acp-21-4599-2021. Liu, J., K. W. Bowman, D. S. Schimel, N. C. Parazoo, Z. Jiang, M. Lee, A. A. Bloom, D. Wunch, C. Frankenberg, Y. Sun, C. W. O'Dell, K. R. Gurney, D. Menemenlis, M. Gierach, D. Crisp, and A. Eldering (2017), Contrasting carbon cycle responses of the tropical continents to the 2015–2016 El Niño, <i>Science</i>, <i>358</i>(6360), eaam5690, doi: 10.1126/science.aam5690. Liu, Z., P. Ciais, Z. Deng, R. Lei, S. J. Davis, S. Feng, B. Zheng, D. Cui, X. Dou, B. Zhu, R. Guo, P. Ke, T. Sun, C. Lu, P. He, Y. Wang, X. Yue, Y. Wang, Y. Lei, H. Zhou, Z. Cai, Y. Wu, R. Guo, T. Han, J. Xue, O. Boucher, E. Boucher, F. Chevallier, K. Tanaka, Y. Wei, H. Zhong, C. Kang, N. Zhang, B. Chen, F. Xi, M. Liu, FM. Bréon, Y. Lu, Q. Zhang, D. Guan, P. Gong, D. M. Kammen, K. He, and H. J. Schellnhuber (2020), Near-real-time monitoring of global CO₂ emissions reveals the effects of the COVID-19 pandemic, <i>Nature Comm.</i>, <i>11</i>(1), 5172, doi:10.1038/s41467-020-18922-7. Lovenduski, N. S., N. C. Swart, A. J. Sutton, J. C. Fyfe, G. A. McKinley, C. Sabine, and N. L. Williams (2021), The ocean carbon response to COVID-related
483 485 486 487 488 489 490 491 492 493 494 493 494 495 496 497 498 499 500 501 502 503 504	 ^{A.} Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson, and M. W. Jones (2021), Fossil CO₂ emissions in the post-COVID-19 era, <i>Nature Clim. Change</i>, <i>11</i>(3), 197–199, doi:10.1038/s41558-021-01001-0. Liu, D., W. Sun, N. Zeng, P. Han, B. Yao, Z. Liu, P. Wang, K. Zheng, H. Mei, and Q. Cai (2021), Observed decreases in on-road CO₂ concentrations in Beijing during COVID-19 restrictions, <i>Atmos. Chem. Phys.</i>, <i>21</i>(6), 4599–4614, doi:10.5194/acp-21-4599-2021. Liu, J., K. W. Bowman, D. S. Schimel, N. C. Parazoo, Z. Jiang, M. Lee, A. A. Bloom, D. Wunch, C. Frankenberg, Y. Sun, C. W. O'Dell, K. R. Gurney, D. Menemenlis, M. Gierach, D. Crisp, and A. Eldering (2017), Contrasting carbon cycle responses of the tropical continents to the 2015–2016 El Niño, <i>Science</i>, <i>358</i>(6360), eaam5690, doi: 10.1126/science.aam5690. Liu, Z., P. Ciais, Z. Deng, R. Lei, S. J. Davis, S. Feng, B. Zheng, D. Cui, X. Dou, B. Zhu, R. Guo, P. Ke, T. Sun, C. Lu, P. He, Y. Wang, X. Yue, Y. Wang, Y. Lei, H. Zhou, Z. Cai, Y. Wu, R. Guo, T. Han, J. Xue, O. Boucher, E. Boucher, F. Chevallier, K. Tanaka, Y. Wei, H. Zhong, C. Kang, N. Zhang, B. Chen, F. Xi, M. Liu, FM. Bréon, Y. Lu, Q. Zhang, D. Guan, P. Gong, D. M. Kammen, K. He, and H. J. Schellnhuber (2020), Near-real-time monitoring of global CO₂ emissions reveals the effects of the COVID-19 pandemic, <i>Nature Comm.</i>, <i>11</i>(1), 5172, doi:10.1038/s41467-020-18922-7. Lovenduski, N. S., N. C. Swart, A. J. Sutton, J. C. Fyfe, G. A. McKinley, C. Sabine, and N. L. Williams (2021), The ocean carbon response to COVID-related emissions reductions, <i>Geophys. Res. Lett.</i>, <i>48</i>(6), e2020GL092,263, doi: http://doi.org/line.com/l
483 484 485 486 487 488 489 490 491 492 493 494 493 494 495 496 497 498 499 500 501 502 503 504 505	 ^{A.} Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson, and M. W. Jones (2021), Fossil CO₂ emissions in the post-COVID-19 era, <i>Nature Clim. Change</i>, <i>11</i>(3), 197–199, doi:10.1038/s41558-021-01001-0. Liu, D., W. Sun, N. Zeng, P. Han, B. Yao, Z. Liu, P. Wang, K. Zheng, H. Mei, and Q. Cai (2021), Observed decreases in on-road CO₂ concentrations in Beijing during COVID-19 restrictions, <i>Atmos. Chem. Phys.</i>, <i>21</i>(6), 4599–4614, doi:10.5194/acp-21-4599-2021. Liu, J., K. W. Bowman, D. S. Schimel, N. C. Parazoo, Z. Jiang, M. Lee, A. A. Bloom, D. Wunch, C. Frankenberg, Y. Sun, C. W. O'Dell, K. R. Gurney, D. Menemenlis, M. Gierach, D. Crisp, and A. Eldering (2017), Contrasting carbon cycle responses of the tropical continents to the 2015–2016 El Niño, <i>Science</i>, <i>358</i>(6360), eaam5690, doi: 10.1126/science.aam5690. Liu, Z., P. Ciais, Z. Deng, R. Lei, S. J. Davis, S. Feng, B. Zheng, D. Cui, X. Dou, B. Zhu, R. Guo, P. Ke, T. Sun, C. Lu, P. He, Y. Wang, X. Yue, Y. Wang, Y. Lei, H. Zhou, Z. Cai, Y. Wu, R. Guo, T. Han, J. Xue, O. Boucher, E. Boucher, F. Chevallier, K. Tanaka, Y. Wei, H. Zhong, C. Kang, N. Zhang, B. Chen, F. Xi, M. Liu, FM. Bréon, Y. Lu, Q. Zhang, D. Guan, P. Gong, D. M. Kammen, K. He, and H. J. Schellnhuber (2020), Near-real-time monitoring of global CO₂ emissions reveals the effects of the COVID-19 pandemic, <i>Nature Comm.</i>, <i>11</i>(1), 5172, doi:10.1038/s41467-020-18922-7. Lovenduski, N. S., N. C. Swart, A. J. Sutton, J. C. Fyfe, G. A. McKinley, C. Sabine, and N. L. Williams (2021), The ocean carbon response to COVID-related emissions reductions, <i>Geophys. Res. Lett.</i>, <i>48</i>(6), e2020GL092,263, doi: https://doi.org/10.1029/2020GL092263.
484 485 486 487 488 489 490 491 492 493 493 493 493 493 493 495 496 497 498 499 500 501 502 503 504 505 506	 ^{A.} Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson, and M. W. Jones (2021), Fossil CO₂ emissions in the post-COVID-19 era, <i>Nature Clim. Change</i>, <i>11</i>(3), 197–199, doi:10.1038/s41558-021-01001-0. Liu, D., W. Sun, N. Zeng, P. Han, B. Yao, Z. Liu, P. Wang, K. Zheng, H. Mei, and Q. Cai (2021), Observed decreases in on-road CO₂ concentrations in Beijing during COVID-19 restrictions, <i>Atmos. Chem. Phys.</i>, <i>21</i>(6), 4599–4614, doi:10.5194/acp-21-4599-2021. Liu, J., K. W. Bowman, D. S. Schimel, N. C. Parazoo, Z. Jiang, M. Lee, A. A. Bloom, D. Wunch, C. Frankenberg, Y. Sun, C. W. O'Dell, K. R. Gurney, D. Menemenlis, M. Gierach, D. Crisp, and A. Eldering (2017), Contrasting carbon cycle responses of the tropical continents to the 2015–2016 El Niño, <i>Science</i>, <i>358</i>(6360), eaam5690, doi: 10.1126/science.aam5690. Liu, Z., P. Ciais, Z. Deng, R. Lei, S. J. Davis, S. Feng, B. Zheng, D. Cui, X. Dou, B. Zhu, R. Guo, P. Ke, T. Sun, C. Lu, P. He, Y. Wang, X. Yue, Y. Wang, Y. Lei, H. Zhou, Z. Cai, Y. Wu, R. Guo, T. Han, J. Xue, O. Boucher, E. Boucher, F. Chevallier, K. Tanaka, Y. Wei, H. Zhong, C. Kang, N. Zhang, B. Chen, F. Xi, M. Liu, FM. Bréon, Y. Lu, Q. Zhang, D. Guan, P. Gong, D. M. Kammen, K. He, and H. J. Schellnhuber (2020), Near-real-time monitoring of global CO₂ emissions reveals the effects of the COVID-19 pandemic, <i>Nature Comm.</i>, <i>11</i>(1), 5172, doi:10.1038/s41467-020-18922-7. Lovenduski, N. S., N. C. Swart, A. J. Sutton, J. C. Fyfe, G. A. McKinley, C. Sabine, and N. L. Williams (2021), The ocean carbon response to COVID-related emissions reductions, <i>Geophys. Res. Lett.</i>, <i>48</i>(6), e2020GL092,263, doi: https://doi.org/10.1029/2020GL092263. McKinley, G. A., A. R. Fay, Y. A. Eddebbar, L. Gloege, and N. S. Lovenduski (2020), Exter-
484 485 486 487 488 489 490 491 492 493 494 493 494 495 496 499 500 501 502 500 501 502 503 504 505 506 506	 ^{A.} Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson, and M. W. Jones (2021), Fossil CO₂ emissions in the post-COVID-19 era, <i>Nature Clim. Change</i>, <i>11</i>(3), 197–199, doi:10.1038/s41558-021-01001-0. Liu, D., W. Sun, N. Zeng, P. Han, B. Yao, Z. Liu, P. Wang, K. Zheng, H. Mei, and Q. Cai (2021), Observed decreases in on-road CO₂ concentrations in Beijing during COVID-19 restrictions, <i>Atmos. Chem. Phys.</i>, <i>21</i>(6), 4599–4614, doi:10.5194/acp-21-4599-2021. Liu, J., K. W. Bowman, D. S. Schimel, N. C. Parazoo, Z. Jiang, M. Lee, A. A. Bloom, D. Wunch, C. Frankenberg, Y. Sun, C. W. O'Dell, K. R. Gurney, D. Menemenlis, M. Gierach, D. Crisp, and A. Eldering (2017), Contrasting carbon cycle responses of the tropical continents to the 2015–2016 El Niño, <i>Science</i>, <i>358</i>(6360), eaam5690, doi: 10.1126/science.aam5690. Liu, Z., P. Ciais, Z. Deng, R. Lei, S. J. Davis, S. Feng, B. Zheng, D. Cui, X. Dou, B. Zhu, R. Guo, P. Ke, T. Sun, C. Lu, P. He, Y. Wang, X. Yue, Y. Wang, Y. Lei, H. Zhou, Z. Cai, Y. Wu, R. Guo, T. Han, J. Xue, O. Boucher, E. Boucher, F. Chevallier, K. Tanaka, Y. Wei, H. Zhong, C. Kang, N. Zhang, B. Chen, F. Xi, M. Liu, FM. Bréon, Y. Lu, Q. Zhang, D. Guan, P. Gong, D. M. Kammen, K. He, and H. J. Schellnhuber (2020), Near-real-time monitoring of global CO₂ emissions reveals the effects of the COVID-19 pandemic, <i>Nature Comm.</i>, <i>11</i>(1), 5172, doi:10.1038/s41467-020-18922-7. Lovenduski, N. S., N. C. Swart, A. J. Sutton, J. C. Fyfe, G. A. McKinley, C. Sabine, and N. L. Williams (2021), The ocean carbon response to COVID-related emissions reductions, <i>Geophys. Res. Lett.</i>, <i>48</i>(6), e2020GL092,263, doi: https://doi.org/10.1029/2020GL092263. McKinley, G. A., A. R. Fay, Y. A. Eddebbar, L. Gloege, and N. S. Lovenduski (2020), External forcing explains recent decadal variability of the ocean carbon sink, <i>AGU Advances</i>, 140.
484 485 486 487 488 489 490 491 492 493 494 495 494 495 496 497 498 499 500 501 502 503 504 505 505 506 507 508	 ^{A.} Le Quéré, C., G. P. Peters, P. Friedlingstein, R. M. Andrew, J. G. Canadell, S. J. Davis, R. B. Jackson, and M. W. Jones (2021), Fossil CO₂ emissions in the post-COVID-19 era, <i>Nature Clim. Change</i>, <i>11</i>(3), 197–199, doi:10.1038/s41558-021-01001-0. Liu, D., W. Sun, N. Zeng, P. Han, B. Yao, Z. Liu, P. Wang, K. Zheng, H. Mei, and Q. Cai (2021), Observed decreases in on-road CO₂ concentrations in Beijing during COVID-19 restrictions, <i>Atmos. Chem. Phys.</i>, <i>21</i>(6), 4599–4614, doi:10.5194/acp-21-4599-2021. Liu, J., K. W. Bowman, D. S. Schimel, N. C. Parazoo, Z. Jiang, M. Lee, A. A. Bloom, D. Wunch, C. Frankenberg, Y. Sun, C. W. O'Dell, K. R. Gurney, D. Menemenlis, M. Gierach, D. Crisp, and A. Eldering (2017), Contrasting carbon cycle responses of the tropical continents to the 2015–2016 El Niño, <i>Science</i>, <i>358</i>(6360), eaam5690, doi: 10.1126/science.aam5690. Liu, Z., P. Ciais, Z. Deng, R. Lei, S. J. Davis, S. Feng, B. Zheng, D. Cui, X. Dou, B. Zhu, R. Guo, P. Ke, T. Sun, C. Lu, P. He, Y. Wang, X. Yue, Y. Wang, Y. Lei, H. Zhou, Z. Cai, Y. Wu, R. Guo, T. Han, J. Xue, O. Boucher, E. Boucher, F. Chevallier, K. Tanaka, Y. Wei, H. Zhong, C. Kang, N. Zhang, B. Chen, F. Xi, M. Liu, FM. Bréon, Y. Lu, Q. Zhang, D. Guan, P. Gong, D. M. Kammen, K. He, and H. J. Schellnhuber (2020), Near-real-time monitoring of global CO₂ emissions reveals the effects of the COVID-19 pandemic, <i>Nature Comm.</i>, <i>11</i>(1), 5172, doi:10.1038/s41467-020-18922-7. Lovenduski, N. S., N. C. Swart, A. J. Sutton, J. C. Fyfe, G. A. McKinley, C. Sabine, and N. L. Williams (2021), The ocean carbon response to COVID-related emissions reductions, <i>Geophys. Res. Lett.</i>, <i>48</i>(6), e2020GL092,263, doi: https://doi.org/10.1029/2020GL09263. McKinley, G. A., A. R. Fay, Y. A. Eddebbar, L. Gloege, and N. S. Lovenduski (2020), External forcing explains recent decadal variability of the ocean carbon sink, <i>AGU Advances</i>, <i>1</i>(2), e2019AV000,149, doi:10.1029/2019AV000149.

A. Alkhaled, P. Rayner, D. J. Jacob, P. Suntharalingam, D. B. A. Jones, A. S. Denning,

511	M. E. Nicholls, S. C. Doney, S. Pawson, H. Boesch, B. J. Connor, I. Y. Fung, D. O'Brien,
512	R. J. Salawitch, S. P. Sander, B. Sen, P. Tans, G. C. Toon, P. O. Wennberg, S. C. Wofsy,
513	Y. L. Yung, and R. M. Law (2007), Precision requirements for space-based data, J. Geo-
514	phys. Res. Atm., 112(D10), doi:https://doi.org/10.1029/2006JD007659.
515	NOAA Global Monitoring Laboratory (2021), Despite pandemic shut-
516	downs, carbon dioxide and methane surged in 2020, NOAA Research News,
517	https://research.noaa.gov/article/ArtMID/587/ArticleID/2742/Despite-pandemic-
518	shutdowns-carbon-dioxide-and-methane-surged-in-2020.
519	O'Neill, B. C., C. Tebaldi, D. P. van Vuuren, V. Eyring, P. Friedlingstein, G. Hurtt, R. Knutti,
520	E. Kriegler, JF. Lamarque, J. Lowe, G. A. Meehl, R. Moss, K. Riahi, and B. M. Sander-
521	son (2016), The Scenario Model Intercomparison Project (ScenarioMIP) for CMIP6,
522	Geosci. Model Dev., 9(9), 3461–3482, doi:10.5194/gmd-9-3461-2016.
523	Peters, G. P., C. Le Quéré, R. M. Andrew, J. G. Canadell, P. Friedlingstein, T. Ilyina, R. B.
524	Jackson, F. Joos, J. I. Korsbakken, G. A. McKinley, S. Sitch, and P. Tans (2017), To-
525	wards real-time verification of CO ₂ emissions, <i>Nature Clim. Change</i> , 7(12), 848–850,
526	doi:10.1038/s41558-017-0013-9.
527	Rayner, P. J., and D. M. O'Brien (2001), The utility of remotely sensed CO ₂ concen-
528	tration data in surface source inversions, Geophys. Res. Lett., 28(1), 175–178, doi:
529	https://doi.org/10.1029/2000GL011912.
530	Ridge, S. M., and G. A. McKinley (2021), Ocean carbon uptake under aggressive emission
531	mitigation, <i>Biogeosciences</i> , 18(8), 2711–2725, doi:10.5194/bg-18-2711-2021.
532	Sarmiento, J. L., and N. Gruber (2002), Sinks for anthropogenic carbon, <i>Phys. Today</i> , 55(8),
533	30–36, doi:10.1063/1.1510279.
534	Sellers, P. J., D. S. Schimel, B. Moore, J. Liu, and A. Eldering (2018), Observing car-
535	bon cycle–climate feedbacks from space, <i>Proc. Nat. Acad. Sci.</i> , 115(31), 7860, doi:
536	10.1073/pnas.1716613115.
537	Sutton, A. J., R. A. Feely, S. Maenner-Jones, S. Musielwicz, J. Osborne, C. Dietrich,
538	N. Monacci, J. Cross, R. Bott, A. Kozyr, A. J. Andersson, N. R. Bates, WJ. Cai, M. F.
539	Cronin, E. H. De Carlo, B. Hales, S. D. Howden, C. M. Lee, D. P. Manzello, M. J.
540	McPhaden, M. Melendez, J. B. Mickett, J. A. Newton, S. E. Noakes, J. H. Noh, S. K.
541	(2010) Autonomous constant (CO) and all time conics from 40 surface busines and
542	(2019), Autonomous seawater pCO_2 and pH time series from 40 surface buoys and the emergence of orthogonagenia trands. Earth Sust. Soi. Data $11(1)$, 421, 420, doi:
543	10 5104/accd 11 421 2010
544	10.5194/CSSU-11-421-2017. Swort N.C. I.N.S. Cole, V.V. Kharin, M. Lazara, I.E. Scinocca, N.D. Gillett, I. Anstev
545	V Arora I P. Christian S. Hanna V. Jiao W.G. Lea F. Majaess, O. A. Saenko
546	C Seiler C Seinen A Shao M Sigmond I Solheim K von Salzen D Vang and
547	B Winter (2019) The Canadian Earth System Model version 5 (CanESM50.3) <i>Geosci</i>
540	Model Dev 12(11) 4823–4873 doi:10.5194/gmd-12-4823-2019
545	Tohiima V P K Patra V Niwa H Mukai M Sasakawa and T Machida (2020) Detection
551	of fossil-fuel CO ₂ plummet in China due to COVID-19 by observation at Hateruma <i>Sci</i>
552	Rep. 10(1), 18.688, doi:10.1038/s41598-020-75763-6.
552	Turner A I I Kim H Fitzmaurice C Newman K Worthington K Chan P I
554	Wooldridge, P. Köehler, C. Frankenberg, and R. C. Cohen (2020). Observed impacts of
555	COVID-19 on urban CO ₂ emissions, <i>Geophys. Res. Lett.</i> , 47(22), e2020GL090.037, doi:
556	https://doi.org/10.1029/2020GL090037.
557	World Meteorological Organization (2020), Can we see the impact of COVID-19 confine-
558	ment measures on CO ₂ levels in the atmosphere?, WMO Greenhouse Gas Bulletin, num-
559	ber 16, 23 November 2020.
560	Wu, S., W. Zhou, X. Xiong, G. S. Burr, P. Cheng, P. Wang, Z. Niu, and Y. Hou (2021), The
561	impact of COVID-19 lockdown on atmospheric CO ₂ in Xi'an, China, <i>Environ. Res.</i> , 197,
562	111,208, doi:https://doi.org/10.1016/j.envres.2021.111208.
563	Yeager, S. G., G. Danabasoglu, N. A. Rosenbloom, W. Strand, S. C. Bates, G. A. Meehl,
564	A. R. Karspeck, K. Lindsay, M. C. Long, H. Teng, and N. S. Lovenduski (2018), Predict-

- ing near-term changes in the Earth system: A large ensemble of initialized decadal predic-
- tion simulations using the Community Earth System Model, B. Am. Meteorol. Soc., 99(9),
- ⁵⁶⁷ 1867–1886, doi:10.1175/BAMS-D-17-0098.1.



Figure S1. Global-mean, de-seasoned CO₂ emissions (Pg C yr⁻¹) (a) over the historical period, and (b) for the (black) control / SSP2-4.5, (red) COVID-like, (green) $2 \times$ COVID-like, and (blue) $4 \times$ COVID-like

570 scenarios.



Figure S2. (a) Spatial distribution of 2020 annual-mean CO₂ emissions for the control (SSP2-4.5) en-

semble (kg m⁻² yr⁻¹). (b) Seasonally varying 2020 global-mean CO₂ emissions for the control (SSP2-4.5)

 $_{573}$ ensemble (Pg C yr⁻¹).



Figure S3. Monthly χ CO₂ anomaly (ppm) relative to the time-mean χ CO₂ from (gray) the CanESM5

- control ensemble and (purple) observations during 2015 at (a) Point Barrow, (b) Mauna Loa, and (c) South
- 576 Pole stations.